Click here to view this article's online features:

ANNUAL Further

- Download figures as PPT slides
 Navigate linked references
- Download citations
- Explore related articles
- Search keywords

Combustion and Engine-Core Noise

Matthias Ihme

Department of Mechanical Engineering, Stanford University, Stanford, California 94305; email: mihme@stanford.edu

Annu. Rev. Fluid Mech. 2017. 49:277-310

First published online as a Review in Advance on August 10, 2016

The Annual Review of Fluid Mechanics is online at fluid.annualreviews.org

This article's doi: 10.1146/annurev-fluid-122414-034542

Copyright © 2017 by Annual Reviews. All rights reserved

Keywords

combustion noise, engine-core noise, jet-exhaust noise, aeroacoustics, computational aeroacoustics

Abstract

The implementation of advanced low-emission aircraft engine technologies and the reduction of noise from airframe, fan, and jet exhaust have made noise contributions from an engine core increasingly important. Therefore, meeting future ambitious noise-reduction goals requires the consideration of engine-core noise. This article reviews progress on the fundamental understanding, experimental analysis, and modeling of engine-core noise; addresses limitations of current techniques; and identifies opportunities for future research. After identifying core-noise contributions from the combustor, turbomachinery, nozzles, and jet exhaust, they are examined in detail. Contributions from direct combustion noise, originating from unsteady combustion, and indirect combustion noise, resulting from the interaction of flow-field perturbations with mean-flow variations in turbine stages and nozzles, are analyzed. A new indirect noise-source contribution arising from mixture inhomogeneities is identified by extending the theory. Although typically omitted in core-noise analysis, the impact of mean-flow variations and nozzle-upstream perturbations on the jet-noise modulation is examined, providing potential avenues for future core-noise mitigation.

1. INTRODUCTION

According to the global air transportation outlook by the International Civil Aviation Organization (ICAO 2013), civil air traffic is projected to grow at an annual rate of 4.6 %, resulting in a doubling of air transportation over the next 15 years. Although the increasing demand in passenger and cargo transportation, constraining factors are operating cost, airport capacities, and environmental pollution and noise. Aircraft noise is of major concern because it adversely affects the quality of life, health, and property value of communities in proximity to airports and main flight corridors. A balanced approach is pursued to mitigate noise exposure by considering residential and land-use planning, operational procedures, operating restrictions, and noise reduction at the source. Whereas the first three strategies are concerned with regulatory noise-mitigation measures, the last strategy is potentially most effective as it aims to reduce noise at its inception.

Main contributors to noise emission from aircraft are the airframe and engines. Airframe noise, typically broadbanded with tonal components, is generated by the fuselage, nacelle, landing gear, wings, and other lifting surfaces. Engine noise has contributions from the fan, compressor, combustor, turbine, and jet exhaust and is examined in this article.

Since the introduction of the jet engine, substantial reductions in engine noise have been achieved. **Figure 1** shows the evolution of the cumulative noise emission of civil aircraft since 1960. Results are obtained from the ICAO noise database (ICAO 2014) and are presented relative to Chapter 3 of the ICAO standard.



Figure 1

Evolution of cumulative aircraft noise reduction at the source (relative to Chapter 3 of International Civil Aviation Organization standard) and future technology goals for commercial aircraft. The symbol size is scaled by the engine bypass ratio, emphasizing the correlation with the noise reduction. Linear regression results are shown by solid gray lines. Data are taken from the ICAO (2014) database. Abbreviation: EPNdB, effective perceived noise level in decibels.

These improvements were primarily achieved through the introduction of the turbofan engine and the successive increase in bypass ratio, thereby reducing contributions from jet noise as a result of the lower jet-exhaust velocity. However, with the reduction in jet noise, core noise together with fan and airframe noise became a leading contributor to overall aircraft noise. The significance of core noise has already been recognized at low-power engine conditions during landing and approach and in auxiliary power units. In addition, the introduction of advanced combustion strategies, such as lean prevaporized premixed combustion, lean direct injection, and other lowemission combustion technologies (Hultgren 2011, Chang et al. 2013), can result in increased noise emissions due to the occurrence of thermoacoustic instabilities, thermal stratification, and higher turbulence levels. Other factors include the consideration of higher overall pressure ratios, higher turbine loadings, and the introduction of engine designs with higher power densities for integration in future aircraft concepts (Mongeau et al. 2013).

The recent resurgence of interest in combustion and engine-core noise is largely stipulated by ambitious aircraft noise-reduction goals (see **Figure 1**) and is enabled by progress on advanced diagnostics, computational modeling, and improved understanding of fundamental physical processes. Progress on this subject has been reviewed previously. For example, Strahle (1978) focused on noise generation from unconfined flames; Candel et al. (2009) examined scaling, theory, and experimental progress on direct combustion noise and flame dynamics in confined geometries; and Dowling & Mahmoudi (2015) analyzed mostly indirect combustion noise.

By complementing these previous reviews, here I provide a holistic discussion of noisegenerating processes induced by unsteady combustion in the engine core. This article expands on the subject of direct and indirect combustion noise by also considering their impact on jet noise and far-field radiation. To this end, Section 2 provides a general overview of core-noise mechanisms and their coupling with external jet noise. This overview serves as a road map for the subsequent component-noise analysis from combustor, nozzle, turbine, and jet exhaust. With relevance to direct combustion noise, Section 3 examines experimental and modeling aspects of noise generation from unsteady combustion. The transmission and generation of noise by temperature, velocity, and mixture inhomogeneities in confined flow paths are discussed in Section 4. Section 5 considers the effects of perturbations generated upstream of the nozzle exit on the modulation of jet noise. Section 6 discusses progress on the system modeling of engine-core noise by integrating findings from component-noise analysis. The articles closes by identifying research opportunities to meet future noise-reduction goals.

2. CORE-NOISE MECHANISMS AND COUPLING PROCESSES

Contributions to engine noise that are directly or indirectly generated upstream of the nozzle exit are commonly referred to as core noise. The importance of core noise was first recognized when reconciling excess levels of low-frequency broadband noise at low jet-exhaust velocities that substantially deviate from Lighthill's quadrupole jet-noise scaling (Hubbard & Lassiter 1953). Through experimental investigations, Bushell (1971) attributed this excess noise to mechanisms in the engine core, which are not present in jet-exhaust flows that are free of upstream disturbances.

Over recent years, different core-noise contributions have been identified (Strahle 1978, Mahan & Karchmer 1995). These distinctions were mainly based on the relative importance of certain noise-generating engine components, mutual interaction between different sources, or noise-source separation based on spectral content. Because core-noise mechanisms are engine specific, the introduction of such prior specifications can have substantial ramifications for the accurate description of core-noise radiation. Therefore, it appears prudent to follow a general approach



Figure 2

Core-noise mechanisms, showing the interaction between different noise-source mechanisms as indicated by the arrows.

by including all noise-source components, arising from the compressor, combustor, turbine, and nozzle, and their impact on jet-exhaust noise.

In describing relevant core-noise contributions, it is instructive to distinguish between mechanisms associated with self-noise and those associated with induced noise. **Figure 2** shows enginecore components and corresponding noise mechanisms.

Here we define self-noise as the direct generation of acoustic pressure fluctuations from an engine component. The noise-generating sources are located within the engine component, and noise is generated in the absence of the coupling with other upstream or downstream engine components through inflow perturbations or acoustic reflections. Relevant self-noise contributions arise from the compressor, combustor, turbine, and nozzle. The primary noise emission from rotating turbomachinery, which includes the compressor and turbine, is tonal noise from the rotor at harmonics of the blade-passing frequency. The interaction of the rotor wake with the stator introduces tonal noise at multiples of the blade-passing frequency with azimuthal order corresponding to the Tyler-Sofrin modes (Tyler & Sofrin 1962). Multiple interaction tones are generated from the interference of the fan wake with the first compressor stator and at internal stages due to wake interference and flow interaction at the rotor-stator separation. Superimposed to the tonal noise is broadband noise, which is generated primarily by the scattering of boundary-layer turbulence at the trailing edge of blades, the impinging of the turbulent rotor wake with the

downstream stator, the separation of the turbulent boundary layer at the blade, and through the interaction of the rotating blade with the turbulent boundary layer. Another contribution, called haystacking, results from the broadening of tonal energy through the interaction with turbulence in engine-external shear layers. The investigation of turbomachinery noise has been the subject of extensive research, as reviewed by Peake & Parry (2012).

The self-noise generated inside the combustor arises from direct combustion noise, which is associated with acoustic perturbations generated by unsteady heat release. Combustion-generated noise is broadbanded, having random phase and peak frequencies typically below 500 Hz (Hassan 1974). Although driven by unsteady heat release, these acoustic pressure perturbations are thought to interact only weakly with the turbulent combustion process (Muthukrishnan et al. 1978). In contrast, thermoacoustic instabilities are large-scale coherent pressure fluctuations (Candel 2002, Dowling & Stow 2003, Culick 2006, Lieuwen 2012), having substantially higher acoustic intensities than combustion noise. These instabilities are caused by the coupling between pressure oscillations, entropy and velocity perturbations, local fluctuations in mixture composition, and unsteady heat release. Thermoacoustic instabilities rely on feedback between the flame and acoustics and are therefore regarded as an induced noise mechanism.

Induced noise is generated by inhomogeneities in the velocity, pressure, temperature, and mixture composition that enter the system or are reflected at downstream engine components. These inhomogeneities trigger instabilities or interact with the mean flow, inducing noise that would otherwise be absent for clean inflow and undisturbed exit conditions. Turbulence and flow-field distortions ingested through the engine intake, or the interaction of the fan wake with the booster stage, are indirect noise source contributions in the compressor section. However, owing to high solidity and multiple stages, the propagation of noise through the compressor has substantial transmission losses. Therefore, only noise and perturbations generated at the last compressor stages warrant consideration as potential noise-inducing mechanisms in the combustor. Perturbations leaving the compressor are further attenuated and dispersed as they propagate through the diffusor and enter the combustor. Although signatures of compressor tones have been observed in combustors operating at high power (Reshotko & Karchmer 1980), broadband perturbations entering the combustor are typically outweighed by turbulent mixing and combustion processes inside the combustor. However, if sufficiently large in amplitude, these external perturbations can trigger instabilities in the combustor and undergo subcritical bifurcation (Burnley & Culick 2000). Furthermore, the conversion of entropy waves into pressure fluctuations downstream of the combustor and the subsequent reflection can couple with the combustor acoustics. This mode conversion was found to contribute to low-frequency thermoacoustic oscillations (i.e., rumble) at low-engine-power settings for gas-turbine combustors with spray atomization (Polifke et al. 2001, Zhu et al. 2001). As such, both mechanisms represent routes for induced-noise modulation and interference of broadband noise and coherent pressure oscillations.

Indirect combustion noise is generated by the interaction of inhomogeneities in velocity, temperature, and possibly also mixture composition with mean-flow and pressure gradients in the turbine and nozzle (Cumpsty & Marble 1977a,b; Marble & Candel 1977). Through this interaction, convective entropy and vorticity modes are converted into acoustic pressure waves that are partially reflected as well as transmitted, and ultimately propagate to the far field.

Apart from the direct transmission of the sound generated in the engine core, perturbations and mean-flow distortions induced by unsteady combustion and confinements can impact flow-field conditions at the nozzle exit, thereby indirectly affecting jet-mixing noise. Specifically, it has been recognized that the nozzle geometry, boundary layers, and turbulent, thermal, and acoustic fluctuations can alter jet-noise radiation (Viswanathan 2004, Zaman 2012). Furthermore, jet noise can be modulated by acoustic excitations (Crighton 1981). As such, perturbations originating

from unsteady combustion and tonal noise generated by turbine stages or thermoacoustic instabilities can modulate broadband jet noise, thus indirectly contributing to far-field radiation.

3. DIRECT COMBUSTION NOISE

This section examines the generation of noise from unsteady combustion. It begins with a physical description and then reviews progress on mathematical modeling that includes the characterization of the turbulent flow field, combustion processes, acoustic source term representation, and resulting noise radiation. This is followed by a discussion of challenges and research opportunities.

3.1. Physical Description of Direct Combustion Noise

The direct generation and transmission of combustion noise have been studied experimentally in canonical open-flame configurations operated under premixed and nonpremixed conditions. In foundational experiments on turbulent premixed flames, Smith & Kilham (1963) examined the principal dependence of noise emission on burner geometry, mass flow rate, and fuel mixture. By relating the acoustic power, $P_{\rm ac}$, to flame speed, $S_{\rm L}$, nozzle diameter $D_{\rm J}$, and jet exit velocity $U_{\rm J}$, they observed a monopole-like scaling of the form

$$P_{\rm ac} \propto \rho S_{\rm L}^l D_{\rm I}^m U_{\rm I}^n,\tag{1}$$

with l = 2, m = 2, and n = 2, whereas dimensional arguments dictate (l + n) = 3 and m = 2. This was explained by representing the flame as an uncorrelated distribution of acoustic sources that oscillate with intensity and frequency related to the local heat release and turbulence fluctuations. Bragg (1963) provided a theoretical explanation for this scaling by relating the volumetric expansion of individual source elements with characteristic dimension proportional to the flame thickness.

Thomas & Williams (1966) experimentally examined the monopole noise characteristics further and measured the noise emission from the combustion of a homogeneous spherical premixed mixture. Their analysis showed a quartic dependence of the acoustic power on the burning velocity, and they attributed the difference from the results of Smith & Kilham (1963) to the interference of acoustic sources in a turbulent flame.

Hurle et al. (1968) recognized that the rate of volume variation can be related to the consumption rate of the reactants, which they estimated from narrowband chemiluminescence measurements of excited radicals of C_2 and CH. These free radicals are formed in the reaction zone and are markers for the heat-release rate. This study considered piloted turbulent premixed ethyleneair flames with different burner diameters. Price et al. (1969) extended these measurements to nonpremixed and spray flames with different fuels. Results supported the monopole noise-source character of jet flames, and good agreement between far-field pressure measurements and calculations from emission intensities was obtained. Shivashankara et al. (1974) performed similar emission measurements on a premixed flame and performed cross-correlation analysis that confirmed the direct dependence of the acoustic pressure on the emission intensity.

Kumar (1976) reported substantial differences in the noise characteristics between premixed and nonpremixed flames, which were attributed to differences in the flame structure, flame length, and turbulence. Specifically, this study showed that noise sources in premixed flames are associated with unsteady fluctuations in the flame surface area, which modifies the heat release in the corrugated and thin-reaction zone regimes. In contrast, the main noise sources in nonpremixed flames are associated with dilatational effects generated by mixing and turbulence-induced strain-rate variations. Strahle (1971, 1973) used the acoustic analogy of Lighthill (1952) to develop scaling relations for premixed and nonpremixed flames, confirming the general monopole behavior and low-frequency combustion-noise characteristics. By considering wrinkled flamelet and distributed reaction zone regimes, he found that Equation 1 with exponents l = 3, m = 3, and n = 2 is in better accordance with acoustic power measurements. Scaling relations for combustion noise from turbulent nonpremixed flames were derived by considering the small-turbulent Damköhler limit. However, this limit was obtained from premixed flame arguments (Peters 2000), and a mixing-controlled high-Damköhler limit appears to be more appropriate. The limit of infinitely fast chemistry was considered by Klein & Kok (1999), who developed an integral equation for sound radiation from nonpremixed flames. The sound spectrum was expressed in terms of the turbulence spectrum of the mixture fraction, and numerical simulations of the reacting flow field were used to infer the shape of the acoustic spectrum.

Kotake & Takamoto (1987) examined the effects of turbulence, nozzle geometry, and variations in the equivalence ratio on the acoustic radiation in premixed flames. From correlation measurements between acoustic pressure fluctuations and the time derivative of the light-emission intensity, they found that noise sources in lean premixed flames are mostly distributed in the upper part of the flame, and the dominant noise sources in fuel-rich flames are located in the region below the maximum flame temperature. Although they observed that the acoustic power is insensitive to the level of inflow turbulence at fuel-rich conditions, they showed that higher turbulence levels can lead to a significant increase in acoustic radiation at fuel-lean conditions. These findings are consistent with measurements by Kilham & Kirmani (1979), who rationalized higher noise emissions at lean and stoichiometric conditions with higher turbulent burning velocities.

Rajaram & Lieuwen (2003, 2009) measured the acoustic spectra of turbulent premixed flames and related it to the heat-release spectrum and transfer function. Multiple burner diameters, jet Reynolds numbers, fuels, and equivalence ratios were considered. Using chemiluminescence emission measurements, they observed that the acoustic source is strongest in the upper part of the flame, and its axial extent was correlated to the peak frequency.

Singh et al. (2004, 2005) performed spectral measurements in nonpremixed and partially premixed flames. The hydrodynamic and thermochemical fields in these flames were previously examined (Meier et al. 2000, Schneider et al. 2003) to provide a comprehensive database for model validation. Comparisons with measurements of nonreacting jets at comparable operating conditions showed that premixed flames exhibit higher noise emissions. With an increasing level of partial premixing, the sound spectrum increases at all frequencies, and the velocity scaling exponent of the overall sound pressure level has a strong dependence on the equivalence ratio.

3.2. Mathematical Description of Direct Combustion Noise

The quantitative description of combustion noise requires the consideration of unsteady combustion, the turbulent flow field, acoustic sources, and the transmission of the sound. The scales of these processes are the characteristic acoustic wavelength λ , characteristic flame length *L*, turbulence integral length scale l_t , Kolmogorov length scale η , and flame thickness δ_F (see **Figure 3**). Application of scaling arguments (Crighton 1992, Peters 2000, Veynante & Vervisch 2002) leads to

$$Ma\lambda \sim \xi L \sim l_t \sim Re_t^{3/4} \eta \sim Re_t^{3/4} Ka^{-1/2} \delta_F, \qquad (2)$$

where Ma is the Mach number, Re_t is the turbulence Reynolds number, and Ka is the turbulent Karlovitz number, which compares the characteristic chemical timescale to the Kolmogorov timescale. The scaling parameter for the flame thickness is $\xi \approx \text{Re}_t Z_{\text{st}}^{-1}$ for turbulent diffusion flames (with Z_{st} the stoichiometric mixture fraction), and $\xi \approx (\text{Re}_t \text{Ka}^2)^{1/4} (1 + \text{Re}_t^{1/2})^{-1}$ for



Characteristic scales in a turbulent jet flame.

turbulent premixed flames. Equation 2 shows that for practical applications combustion noise involves a wide range of scales. The separation of these scales depends on the spatial structure of the underlying turbulent flow field, the reaction chemistry and flame structure, and the correlation and spatial extent of the acoustic sources. This section reviews models for these physical mechanisms. Beginning with the mathematical modeling of combusting flows, the description of acoustic sources and noise radiation is then considered.

3.2.1. Mathematical representation of unsteady combusting flows. The spatiotemporal evolution of a chemically reacting flow is described by the conservation equations for mass, momentum, species, and sensible enthalpy (Williams 1985):

$$D_t \rho = -\rho \nabla \cdot \mathbf{u},\tag{3a}$$

$$\rho D_t \mathbf{u} = -\nabla p + \nabla \cdot \boldsymbol{\sigma},\tag{3b}$$

$$\rho D_t \mathbf{Y} = -\nabla \cdot \mathbf{j} + \rho \dot{\boldsymbol{\omega}},\tag{3c}$$

$$\rho D_t b = -\nabla \cdot \mathbf{q} + D_t p + \boldsymbol{\sigma} : \nabla \mathbf{u} + \rho \dot{\omega}_T, \qquad (3d)$$

where $D_t = \partial_t + \mathbf{u} \cdot \nabla$ is the substantial derivative, ρ is the density, \mathbf{u} is the velocity vector, p is the pressure, \mathbf{Y} is the vector of N_s species mass fractions, $\boldsymbol{\omega}$ is the vector of species production rates, b is the sensible enthalpy, and $\dot{\omega}_T$ is the heat-release rate. The viscous stress tensor is denoted by

 σ , **j** is the species diffusion flux matrix, and **q** is the energy flux. Equations 3a–d are accompanied by the ideal gas law, relating the pressure to the density, temperature, and composition:

$$p = \rho RT, \qquad (4)$$

where R is the gas constant, which is a function of the molecular weight W and species composition.

Because of the large number of species, the chemical stiffness, and the scale separation, a direct numerical solution of Equation 3 becomes prohibitive for practical applications. Lowerdimensional manifold representations were developed to reduce the dimensional complexity of the chemistry. In these models, the thermochemical state space is represented in terms of a reduced set of scalars ψ , typically consisting of some subset of mixture fraction Z, reaction progress variable C, strain rate a, and scalar dissipation χ . These manifolds are constructed from the prior solution of representative flame configurations, such as laminar counterflow diffusion flames, freely propagating premixed flames, or unsteady ignition flame elements. Examples of low-dimensional manifold models are the steady laminar flamelet formulation (Peters 1984), flame prolongation of intrinsic lower-dimensional manifold (Gicquel et al. 2000), flamelet-generated manifold method (van Oijen & de Goey 2000), and flamelet/progress variable formulation (Pierce & Moin 2004, Ihme et al. 2005). The last three models share similarities in that the flame structure is obtained from the solution of one-dimensional flamelet equations, and all thermochemical quantities, denoted by the state vector $\mathbf{\Phi} = (\rho, \dot{\boldsymbol{\omega}}, \mathbf{Y}^T, T, \dots)^T$, are parameterized by Z and C. The evolution of the reacting flow field is then described by Equations 3a and 3b and conservation equations for the mixture fraction and progress variable:

$$\rho D_t Z = -\nabla \cdot \mathbf{j}_Z,\tag{5a}$$

$$\rho D_t C = -\nabla \cdot \mathbf{j}_C + \rho \dot{\omega}_C, \tag{5b}$$

and Equation 4 is replaced by a thermochemical state relation:

$$\mathbf{\Phi} = \mathbf{\Phi}(Z, C). \tag{6}$$

Apart from these reaction-transport manifold approaches, other combustion models have been developed, including the conditional moment closure model (Klimenko & Bilger 1999), transported probability density function methods (Pope 1985, Haworth 2010), linear eddy model (Kerstein 1988), and thickened flame model (Colin et al. 2000). Recently, Wu et al. (2015) developed a Pareto-efficient combustion framework that integrates different combustion submodels to provide direct control over model errors and computational complexity subject to user-specific requirements.

In this context, it is important to emphasize that these combustion models rely on different model approximations, which include assumptions about the flame structure, description of scalar mixing processes, and turbulence/chemistry interaction. Although the accuracy of these combustion models has been assessed using direct numerical simulation (DNS) databases and experiments, these comparisons are largely limited to single-point statistics and conditional data of combustion properties. However, acoustic sources typically depend on two-point correlations and integral scales that are rarely considered, except in a few studies (Renfro et al. 2004, Ihme & Pitsch 2012). Because the spatiotemporal correlation of the heat-release rate is related to the acoustic emission, further research is needed to assess combustion models with regard to its accurate description of the coherent flame structure, the spectral content of the acoustic source terms, and transient extinction effects, which all contribute to broadband noise emission.

3.2.2. Mathematical modeling of direct combustion noise. Different methods can be utilized to predict noise emission from reacting flows. Several methods developed for aeroacoustics

applications can be extended to reacting flows, and these methods can be categorized into direct and hybrid techniques (Wang et al. 2006). In the following, both techniques are reviewed with relevance to applications to combustion noise.

3.2.2.1. Direct methods. Direct methods utilize the same physical model to describe the sound generation from the unsteady reacting flow and its propagation, refraction, and scattering. The main advantage of these methods is that they introduce minimal model approximations and explicitly retain the coupling between heat release, species conversion, hydrodynamics, and acoustics. However, to do this, direct methods must represent a wide range of spatial, temporal, and energetic scales, as shown in Equation 2, thus introducing challenges regarding numerical discretization, boundary conditions, spatial resolution, and representation of transport properties and reaction chemistry. Although aspects of the numerical discretization and specification of boundary conditions are common to those employed in aeroacoustics (Wang et al. 2006), others are germane to combustion noise and require special consideration. In particular, the extended combustion domain and the fact that the acoustic sources occupy an extended region of the flame introduce conflicting demands on resolving and containing the entire flame in the computational domain. This contrasts with combustion simulations, in which the spatial resolution is concentrated in regions of high shear, turbulence/chemistry interaction, and chemical nonequilibrium effects near the nozzle exit. The challenge is compounded by the low frequency of the sound, so that simulations of noise propagation over a few characteristic acoustic wavelengths can substantially increase the computational domain and mesh requirements. Furthermore, variations in thermochemical properties in the flame cause acoustic refractions and distortions (Atvars et al. 1966).

Direct computations have been performed to examine combustion-noise sources and to evaluate acoustic analogies. Zhao & Frankel (2001) conducted DNS of an axisymmetric premixed jet, showing that the acoustic source term is noncompact and that the low-frequency noise characteristic is attributed to the shift of shear-layer instabilities to lower frequencies. Talei et al. (2013) used DNS to examine the effects of thermospecies diffusion in an acoustically excited premixed jet flame, showing that annihilation by flame pinch-off and consumption of pockets of unburned mixture are the main monopole sound sources for conditions in which the Lewis number, comparing thermal to mass diffusivity, exceeds unity. This can be explained by the enhanced burning rate as a result of the compressive strain and imbalance between diffusive and thermal transport at the flame tip.

3.2.2.2. Hybrid methods. Hybrid methods employ different solution techniques to represent the unsteady flow field and acoustics. Approaches based on acoustic analogies, scale separation, and flow-field decomposition developed for aeroacoustic applications (Wang et al. 2006) can be extended to combustion noise, and Bailly et al. (2010) provided derivations of acoustic analogies applied to reacting flows.

Acoustic analogies are derived by separating the unsteady flow into a nominal acoustic source and an acoustic propagation operator. Common to all analogies is the assumption that the sound generation is a passive process, and any feedback of the acoustics on the unsteady flow field appears within the source. Different analogies have been developed depending on the decomposition and physical representation of the source and propagation operators.

Perhaps the most prominent acoustic analogy is due to Lighthill (1952), which was originally developed for aeroacoustics and later extended to combustion noise (Crighton et al. 1992, chapter 13). Lighthill's analogy is derived by rearranging the conservation equations for mass and momentum into a linear wave operator for a homogeneous, stationary medium and an equivalent acoustic source term distribution. Using pressure as an aeroacoustic variable, one obtains the pressure form of Lighthill's analogy by performing the operation ∂_t (Equation 3a) $-\nabla \cdot$ (Equation 3b), and adding the term $c_{\infty}^{-2} \partial_t^2 p$ to both sides of the resulting equation (Doak 1972):

$$\left(\boldsymbol{c}_{\infty}^{-2}\partial_{t}^{2}-\nabla^{2}\right)\boldsymbol{p}'=\nabla\cdot\nabla\cdot\left(\boldsymbol{\rho}\mathbf{u}\otimes\mathbf{u}-\boldsymbol{\sigma}\right)-\partial_{t}^{2}(\boldsymbol{\rho}'-\boldsymbol{c}_{\infty}^{-2}\boldsymbol{p}'),\tag{7}$$

where $p' = p - p_{\infty}$ and $\rho' = \rho - \rho_{\infty}$ are pressure and density fluctuations, which are evaluated with respect to the ambient conditions p_{∞} and ρ_{∞} , and c_{∞} is the speed of sound. The first term on the right-hand side combines the unsteady momentum flux and viscous stress tensor, and the last term contains the excess density (Morfey 1973), $\rho_e = \rho' - c_{\infty}^{-2}p'$, which is associated with variations in entropy. Although the excess density is commonly neglected in aeroacoustic applications (Lighthill 1952), unsteady combustion processes introduce substantial entropy fluctuations so that this term becomes an important source for combustion noise. Crighton et al. (1992, chapter 13) rewrote the temporal derivative of the excess density as $\partial_t \rho_e = (\rho_{\infty}/\rho)D_t\rho + \nabla \cdot ([\rho_{\infty} - \rho]\mathbf{u}) - c_{\infty}^{-2}\partial_t p'$, and the Gibbs-Duhem relation was introduced to express the substantial derivative of the density in terms of spatiotemporal variations of the thermochemistry, which arises from diffusive-dissipative processes, heat release, and species conversion.

Flemming et al. (2007) used Lighthill's analogy to predict combustion noise from an open jet diffusion flame. The unsteady turbulent flow field was calculated from a large-eddy simulation (LES) using a mixture-fraction-based flamelet formulation. The acoustic field was computed from a first-order splitting of Lighthill's wave equation, and the second temporal derivative of density was considered as the primary acoustic source.

Ihme et al. (2009) directly coupled a flamelet-progress variable combustion model with Lighthill's acoustic analogy, and the state relation (Equation 6) was used to express the substantial derivative of density in terms of the mixture fraction and progress variable, $D_t \rho = \partial_Z \rho D_t Z + \partial_C \rho D_t C$, resulting in the following equation:

$$\left(c_{\infty}^{-2} \partial_{t}^{2} - \nabla^{2} \right) p' = \nabla \cdot \nabla \cdot (\rho \mathbf{u} \otimes \mathbf{u}) - \partial_{t} \nabla \cdot ([\rho_{\infty} - \rho] \mathbf{u}) - \rho_{\infty} \partial_{t} (\partial_{C} \ln(\rho) \dot{\omega}_{C}) - \rho_{\infty} \partial_{t} \left(\partial_{Z} \rho^{-1} \nabla \cdot \boldsymbol{j}_{Z} + \partial_{C} \rho^{-1} \nabla \cdot \boldsymbol{j}_{C} \right) - \nabla \cdot \nabla \cdot \boldsymbol{\sigma} + c_{\infty}^{-2} \partial_{t}^{2} p',$$
(8)

where the equivalent acoustic sources on the right-hand side represent, respectively, contributions due to Reynolds stresses, the mass flux, reaction rate, diffusive transport in composition space, viscous stresses, and pressure perturbations. Ihme & Pitsch (2012) performed a scaling analysis to show that the far-field acoustic power due to the unsteady Reynolds stresses follows Lighthill's well-known Ma⁵ scaling, the acoustic power by the unsteady mass flux scales with Ma³, and the chemical monopole source term is most prominent. The acoustic power of the chemical source term scales linearly with the Mach number and quadratically with the Damköhler number. Equation 8 was solved by including source terms due to Reynolds stresses, unsteady mass flux, and the chemical source term. These source terms were evaluated from Favre-filtered LES results, and the acoustic far field was evaluated using a free-space Green's function that was solved in Fourier space. Figure 4 illustrates results from this hybrid simulation applied to a turbulent jet diffusion flame (Ihme et al. 2009, Ihme & Pitsch 2012). A comparison of the two acoustic source terms (Figure 4b,c) shows that the source due to the Reynolds stresses is mainly confined to the shear-layer region in the lower part of the flame. In contrast, the source arising from the chemical reaction is aligned with the location of the stoichiometric mixture, extending to the upper part of the flame. Comparisons of the sound pressure level and directivity (Figure 4d,e) confirm that the chemical source term is the dominant source, exhibiting only mild directivity.



Figure 4

Hybrid simulation of combustion noise from a turbulent jet diffusion flame, showing (*a*) the instantaneous temperature field; (*b*) normalized acoustic source term (*color*) and pressure field (*gray*) from unsteady Reynolds stresses, and (*c*) unsteady reaction rate. Source-term contributions to the acoustic far-field pressure are shown for the sound pressure level at (*d*) the 90° sideline and (*e*) directivity. The temperature and acoustic source terms are computed from a large-eddy simulation, and the pressure field is obtained from the integral form of Equation 8. The black lines in panels a-c denote the isocontour of the stoichiometric mixture. Acoustic source terms and pressure in panels *b* and *c* are nondimensionalized by reference quantities and hydrodynamic scales. Figure adapted from lhme et al. (2009) with permission from Elsevier.

Lighthill's wave operator does not include the interaction of the sound by flow-field convection and refraction due to local variations in the speed of sound (Goldstein 1976). Phillips (1960) derived an acoustic analogy accounting for effects of convection and refraction by the flow field and inhomogeneities in the media, which was later extended to combustion noise by Chiu & Summerfield (1974) and Kotake (1975). Phillips's analogy is obtained by combining Equations 3a and 3b and differentiating Equation 4 to express the sensible enthalpy in terms of pressure and density. The resulting inhomogeneous-media wave equation takes the following form:

$$D_t \left(\frac{1}{\gamma} D_t \ln p\right) - \nabla \cdot \left(\frac{c^2}{\gamma} \nabla \ln p\right) = \nabla \mathbf{u} : \nabla \mathbf{u} + D_t \left(\frac{\gamma - 1}{c^2} \dot{\omega}_T\right) + D_t^2 (\ln R) - D_t \left(\frac{\gamma - 1}{\rho c^2} \left[\nabla \cdot \mathbf{q} - \boldsymbol{\sigma} : \nabla \mathbf{u}\right]\right) - \nabla \cdot \left(\rho^{-1} \nabla \cdot \boldsymbol{\sigma}\right), \quad (9)$$

where the left-hand side represents an approximate moving-media wave operator, accounting for effects of convection and refraction of the sound by mean flow and variations of the speed of sound in the source region. However, this incomplete wave operator does not consider effects of the shear on sound propagation (Pridmore-Brown 1958, Doak 1972, Lilley 1974). This shear-refraction term appears as an apparent source as the first term on the right-hand side of Equation 9. Equivalent acoustic source terms on the right-hand side consist of the convective derivative of the heat-release rate, variations in the molecular weight of the gas mixture, and thermoviscous-dissipative effects by energy flux, viscous dissipation, and viscous stresses. Simplifying assumptions commonly invoked to make Phillips's analogy tractable include the consideration of low-speed flows so that substantial derivatives reduce to temporal derivatives, the assumption of a constant specific heat ratio, the consideration of small pressure perturbations, and the omission of viscous-diffusive source terms. In a hybrid simulation, Ihme et al. (2006) employed Phillips's analogy with these approximations to examine effects of the inhomogeneous medium on the directivity pattern from a turbulent jet flame.

The analogy due to Lilley (1974) includes additional propagation effects into the wave operator. This third-order wave equation is obtained by taking the substantial derivative of Equation 9, and introducing the momentum equation to rearrange terms associated with pressure perturbations into the wave propagation operator. Although Lilley's equation has been examined in the aeroacoustics community, it has not yet been applied to combustion noise.

Alternatives to acoustic analogies are hydrodynamic/acoustic splitting techniques (Hardin & Pope 1994) by which the flow field is separated into an incompressible, nonradiating solution that describes the source evolution, and the acoustic field is obtained as a solution to an unsteady compressible perturbation equation. Goldstein (2003) formulated a general splitting theory by separating the flow field into a base flow and a residual flow, resulting in a set of linearized Navier-Stokes equations with a generalized source term. This approach is different from previously discussed analogies that describe the noise radiation by a scalar wave equation for an acoustic variable (pressure or density) and an equivalent source term. Goldstein's formalism was later extended by Seo & Moon (2006) to the linearized perturbed compressible equations to address stability issues, associated with the under-resolved vorticity field.

Ewert & Schröder (2003) considered a different splitting approach and developed the acoustic perturbation equations, and this approach was extended by Bui et al. (2009) to reacting flows. Validation was performed in a jet flame for which the base flow was obtained from a time-averaged LES solution. In this study, the substantial derivative of density was identified as the primary acoustic source term.

3.3. Discussion and Research Needs

Although recent efforts have improved our understanding of direct combustion noise, aspects of source-term identification, mathematical modeling, and applicability of acoustic analogies for predicting combustion noise from confined flames remain open.

With regard to physical understanding, a main ambiguity is the characterization of acoustic source mechanisms from premixed, diffusion, and partially premixed flames. A hypothesis for the

significantly higher noise emission observed in partially premixed flames (Singh et al. 2005) is the presence of a double reaction layer, consisting of a rich premixed zone and a diffusion zone. Another issue is assessing the importance of local extinction and reignition on the acoustic radiation. The frequency of these local quenching and extinction events is related to the excursion of the dissipation rate from its critical value and can contribute to high-frequency noise emissions. Other fundamental aspects are the examination of the effects of refraction and scattering by velocity and temperature gradients in the flames. Although viscous-diffusive contributions are commonly regarded as acoustically inefficient (Obermeier 1985), they can potentially become important in reacting flows (Morfey 2003); this issue has not yet been fully addressed and requires further analysis.

In the context of turbulent combustion modeling, contributions arising from combustionmodel approximations, subgrid-scale models, and turbulent/chemistry closures to noise radiation have not been fully examined. In simulations of practical configurations, the flame structure is not fully resolved over the extended source domain so that closures for nonlinear acoustic sources and turbulence/chemistry interaction may require consideration for the accurate description of far-field radiation. DNS can assist in assessing the importance of these sound source contributions and developing combustion-noise closure models.

Although acoustic analogies and hybrid methods have been successfully applied to combustion noise in open flames, the one-way coupling introduces limitations in applications to direct combustion noise of confined flames. Specifically, the interaction of the sound with the hydrodynamic flow through the scattering and reflection of the pressure at walls and nozzles requires consideration to describe precursors that trigger thermoacoustic instabilities. Multiple-scale techniques, recently developed for the two-way linear coupling of reactive hydrodynamics and acoustics (Magri et al. 2017), represent a promising avenue for application to combustion noise. A higher level of fidelity is achievable by solving the compressible reacting flow equations through LES in conjunction with suitable combustion models. Although this approach represents the nonlinear coupling, the low–Mach number operating condition in gas-turbine combustors imposes severe time-step constraints.

4. GENERATION AND TRANSMISSION OF NOISE IN NOZZLES AND TURBINE STAGES

Whereas noise directly generated inside the combustion chamber propagates through the combustor-downstream engine components before radiating to the far field, nonacoustic perturbations exiting the combustor represent a potent source of indirect sound. Specifically, turbulent mixing, unsteady combustion, and dilution introduce inhomogeneities in the velocity, temperature, and mixture composition that exit the combustor. Combustors are designed to ensure that fuel conversion is completed inside the combustion chamber to maximize fuel utilization and thermal efficiency. Consequently, the direct noise source is no longer active downstream of the combustor, and the downstream flow can-to a good approximation-be represented as chemically frozen or equilibrated. The interaction of these inhomogeneities with mean-flow gradients encountered in downstream nozzle and turbine stages generates acoustic pressure fluctuations. These noisegenerating mechanisms are collectively referred to as indirect combustion noise. Contributions arising from vorticity fluctuations are associated with vortex noise, and indirect noise generation by temperature fluctuations in the form of hot or cold spots exiting the combustor are referred to as entropy noise. The role of entropy variations as indirect noise contributors was first identified theoretically in the seminal work of Candel (1972) and has been further investigated experimentally and computationally.

This section examines indirect combustion noise. Theoretical analysis tools are first reviewed and then extended to include compositional variations, thereby introducing composition noise as an additional source of indirect combustion noise. Subsequently, these theoretical results are connected to experimental analysis and future research needs are discussed.

4.1. Mathematical Description

The interaction of acoustic and entropy waves with a nonuniform mean flow, encountered in diffusors, nozzles, and turbine stages, generates additional sound that contributes to engine-core noise. Different methods to describe the acoustic transmission and reflection in nozzles and turbines have been developed, including compact nozzle theory (Tsien 1952; Candel 1972; Cumpsty & Marble 1977a,b; Marble & Candel 1977; Dowling 1995), the effective nozzle length method (Stow et al. 2002, Goh & Morgans 2011), the linear nozzle element technique (Moase et al. 2007, Giauque et al. 2012), expansion methods (Mani 1981, Duran & Moreau 2013, Duran & Morgans 2015), nonlinear analyses (Bloy 1979, Huet & Giauque 2013), and numerical simulations (Mühlbauer et al. 2009, Leyko et al. 2011, Palies et al. 2011, Mishra & Bodony 2013, Papadogiannis et al. 2015). Common to all methods is the consideration of a homogeneous gas mixture that is described by conservation of mass, momentum, energy, and entropy.

Although interactions among entropy, acoustic, and vorticity perturbations have been examined (Chu & Kovásznay 1958), the relevance of mixture inhomogeneities as a contributor to indirect combustion noise has not yet been considered. To address this, we examine whether compositional inhomogeneities exiting the combustor as a result of incomplete mixing, mixture stratification, or dilution represent an additional contribution to indirect combustion noise. These mixture inhomogeneities can become increasingly important with the implementation of compact burners, high-power-density engine cores, and advanced low-emission combustors (Hultgren 2011, Chang et al. 2013).

Candel (1972) and Marble & Candel (1977) used compact nozzle theory to examine the transmission and reflection of acoustic and entropy waves in a one-dimensional nozzle. This theory is valid in the limit of small Helmholtz numbers, $\text{He} = l\omega/c$ (with *l* the characteristic length of the nozzle; the speed of sound *c* is commonly evaluated at the nozzle throat), so that perturbations are assumed to propagate quasi-steadily through the nozzle. Using this theory, they derived transfer functions relating acoustic and entropy perturbations between the inlet and outlet of the nozzle. Here, this theory is extended to examine the effects of mixture fluctuations on the transfer matrix. For this, a chemically frozen flow with frozen internal energy modes is considered, so that all species are uniquely represented in terms of the mixture fraction, $\mathbf{Y} = \mathbf{Y}(Z)$.

Governing equations for perturbations of the mass flow rate, total sensible enthalpy, entropy, and mixture fraction (neglecting vortical disturbances) provide jump conditions across the nozzle:

$$\llbracket dm \rrbracket_a^b = 0, \qquad \llbracket db_t \rrbracket_a^b = 0, \qquad \llbracket ds \rrbracket_a^b = 0, \qquad \llbracket dZ \rrbracket_a^b = 0, \tag{10}$$

where the indices *a* and *b* denote the conditions at the inlet and at the exit of the nozzle, respectively (see **Figure 5**).

The jump conditions for the conserved quantities can be related to the primitive state variables by recalling the differentials of the total sensible enthalpy, state equation, and Gibbs equation for a multicomponent gas mixture:

$$\mathrm{d}b_{\mathrm{t}} = \mathrm{d}b + u\mathrm{d}u,\tag{11a}$$

$$dp = RT d\rho + \rho T \frac{dR}{dZ} dZ + \rho R dT, \qquad (11b)$$

$$\mathrm{d}s = \frac{1}{T}\mathrm{d}b - \frac{R}{p}\mathrm{d}p - \frac{1}{T}\sum_{i=1}^{N_s}\frac{\mu_i}{W_i}\frac{\mathrm{d}Y_i}{\mathrm{d}Z}\mathrm{d}Z,\tag{11c}$$



Figure 5

Schematic illustration of a compact nozzle, showing acoustic, entropy, and compositional waves at subsonic conditions. The indices a and b denote the conditions at the inlet and at the exit of the nozzle, respectively.

where $db = c_p dT + \sum_{i=1}^{N_s} b_i (dY_i/dZ) dZ$ and $\mu_i = \mu_i^0 + \mathcal{R}T \ln(p_i/p_0)$ is the chemical potential of species *i*, which is a function of pressure and temperature (Job & Herrmann 2006).

By considering a subcritical nozzle, in which the flow remains subsonic, one can identify four invariants at each side of the nozzle (see **Figure 5**), corresponding to the downstream and upstream propagating acoustic waves, convective entropy wave, and convective compositional wave:

$$\pi^{\pm} = \frac{1}{2} \left(\frac{p'}{\gamma \overline{p}} \pm \frac{u'}{\overline{c}} \right), \qquad \sigma = \frac{s'}{c_p}, \qquad \xi = \frac{1}{R} \frac{\mathrm{d}R}{\mathrm{d}Z} Z'. \tag{12}$$

Expressed in terms of these characteristic variables, the dimensionless form of the conserved quantities can be written as

$$\frac{\dot{m}'}{\ddot{m}} = \left(1 + \frac{1}{\mathrm{Ma}}\right)\pi^+ + \left(1 - \frac{1}{\mathrm{Ma}}\right)\pi^- - \sigma - (\Phi - \Psi + 1)\xi,$$
(13a)

$$\frac{b'_{t}}{\overline{b}_{t}} = \frac{2(\gamma - 1)}{2 + (\gamma - 1)Ma^{2}} \left((1 + Ma)\pi^{+} + (1 - Ma)\pi^{-} + \frac{\sigma + \Phi\xi}{\gamma - 1} \right),$$
(13b)

$$\frac{s'}{c_p} = \sigma, \tag{13c}$$

$$\frac{1}{R}\frac{\mathrm{d}R}{\mathrm{d}Z}Z' = \xi,\tag{13d}$$

where $\Phi = 1/(c_p T) \sum_{i=1}^{N_s} (\mu_i/W_i) (dY_i/dZ) [R/(dR/dZ)]$ and $\Psi = \sum_{i=1}^{N_s} [b_i/(c_p T)] (dY_i/dZ) [R/(dR/dZ)]$. Upon inserting Equations 13a–d into Equation 10, one obtains a linear set of algebraic equations that relates the four incoming waves $(\pi_a^+, \sigma_a, \xi_a, \text{ and } \pi_b^-)$ to the four outgoing waves $(\pi_a^-, \sigma_b, \xi_b, \text{ and } \pi_b^+)$. By setting π_b^- to zero, one can derive transfer functions for the acoustic response at the nozzle exit to inlet excitations by acoustic, entropy, and compositional perturbations. These relations are functions of the Mach number and the chemical potential at the inlet and exit of the nozzle, taking the following form:

$$\frac{\pi_b^+}{\pi_a^+} = \frac{2(1 + Ma_a)Ma_b}{(1 + Ma_b)(Ma_a + Ma_b)} \frac{\left(1 + \frac{\gamma - 1}{2}Ma_b^2\right)}{\left(1 + \frac{\gamma - 1}{2}Ma_aMa_b\right)},$$
(14a)

$$\frac{\pi_b^+}{\sigma_a} = \frac{(Ma_b - Ma_a)Ma_b}{2(1 + Ma_b)\left(1 + \frac{\gamma - 1}{2}Ma_aMa_b\right)},$$
(14b)

$$\frac{\pi_b^+}{\xi_a} = \frac{(\gamma - 1)(\Phi_b - \Phi_a) \left(1 + \frac{\gamma - 1}{2} M a_b^2\right) M a_a M a_b}{(\gamma - 1)(1 + M a_b)(M a_a + M a_b) \left(1 + \frac{\gamma - 1}{2} M a_a M a_b\right)} + \frac{M a_b \left[(\Phi_a - \Phi_b) + \frac{(\gamma - 1)}{2} (\Phi_a M a_b^2 - \Phi_b M a_a^2)\right]}{(\gamma - 1)(1 + M a_b)(M a_a + M a_b) \left(1 + \frac{\gamma - 1}{2} M a_a M a_b\right)},$$
(14c)

and similar relations for the reflected waves π_a^- can be derived. The transfer functions given in Equations 14a and 14b were derived by Marble & Candel (1977), and Equation 14c introduces an additional indirect noise contribution that is caused by nozzle-upstream perturbations in the mixture composition. With the assumption of equal chemical potential and thermochemical state at both sides of the nozzle, $\Phi_a = \Phi_b$, the ratio between the indirect mixing noise and entropy noise can be computed as

$$\frac{\pi_b^+}{\xi_a}\frac{\sigma_a}{\pi_b^+} = \Phi, \tag{15}$$

which depends on the chemical potential of the mixture composition. A similar analysis can be performed for choked and supersonic nozzles, where additional conditions for the critical mass flow and shock relations are required to close the system of linear equations.

Several assumptions on the state of the mixture were introduced in the above derivation to obtain closed-form transfer functions. Further analysis is necessary to quantify the significance of the herein identified mixing-induced indirect combustion noise.

Owing to intrinsic assumptions, the application of compact nozzle theory is restricted to conditions that are described by small Helmholtz numbers, which are often adequate for providing reasonable estimations of indirect combustion noise (Stow et al. 2002, Leyko et al. 2009). However, extensions of this theory are necessary to consider effects of nozzle geometry and finite disturbances. In fact, Marble & Candel (1977) considered the transmission and reflection in a choked nozzle of finite length. For this, they assumed a linear velocity profile through the nozzle, resulting in a hypergeometric differential equation. From the solution, they showed that the amplitude and phase of the transmitted acoustic waves exhibit a pronounced dependence on the Helmholtz number and nozzle-exit Mach number. By approximating the nozzle by elements with piecewise linear velocity profiles and appropriate matching conditions, one can further extend this theory to represent more complex geometries. This was done by Moase et al. (2007), who considered the forced frequency response of choked nozzles and supersonic diffusors of arbitrary shape. Through this analysis, the nonlinear nozzle response was shown to become important with increasing Mach number and forcing amplitude. Later, Giauque et al. (2012) applied this linear nozzle-element model to subcritical nozzles to examine the influence of nozzle geometry on the transfer function.

Instead of representing the nozzle by elements with linear velocity profiles, Stow et al. (2002) considered a different approach in which the perturbed flow solution was expanded for low frequencies: $\phi'(x, \text{He}) = \phi'_0(x) + \text{He}\phi'_1(x) + \mathcal{O}(\text{He}^2)$. The method was applied to annular flows in choked nozzles. The zeroth-order term recovered the reflection coefficients for compact choked nozzles, and the first-order correction introduces velocity integrals, which were evaluated by approximating the nozzle by a straight duct with an effective length represented by the inlet mean velocity and the convective time through the throat. This first-order term provides a phase correction to the reflection coefficient. Goh & Morgans (2011) extended this analysis to examine the response of supercritical nozzles to acoustic and entropy disturbances.

Duran & Moreau (2013) studied transfer functions of subsonic and choked nozzles by formulating the linearized Euler equations in terms of the invariants given in Equations 13a–c, resulting in a system of variable-coefficient linear differential equations. Instead of directly expanding the solution vector, they used the Magnus expansion (Blanes et al. 2009) to expand the coefficient matrix. This approach allows the consideration of general continuous mean-flow profiles and nozzle geometries. The method was applied to compute reflection and transmission coefficients of subsonic and choked nozzles and was later extended by Duran & Morgans (2015) to examine azimuthal waves in annular nozzles. Results obtained with this method support previous findings that transfer functions exhibit a strong dependence on perturbation frequency and nozzle geometry. In addition to extending the analysis to finite frequency response, the nonlinear perturbation response was addressed by Huet & Giauque (2013). They considered a quasi-steady flow evolution in order to obtain an analytical evaluation of the Riemann invariants in the presence of large-amplitude entropy excitations.

The generation of indirect noise by entropy disturbances in a turbine cascade was first studied by Cumpsty & Marble (1977a,b). In this analysis, they assumed that the blade row was compact and so could be represented by an infinitely thin disk. Application to isolated blade rows and compressor stages showed that entropy noise increases with increasing Mach number, pressure ratio, and blade loading, which provides an explanation for observed excess noise from gas turbines. Compared to recent advances on the formulation of analytic methods for nozzle flows, extensions to finite-frequency response and the description of stage geometries have received only little attention; further research is therefore needed on developing transfer functions with higher physical fidelity for rotating turbomachinery.

Numerical simulations were performed to evaluate low-order models and examine transfer mechanisms. Leyko et al. (2010) and Mishra & Bodony (2013) evaluated the accuracy of actuator disk theory by Cumpsty & Marble (1977b) through two-dimensional numerical simulations. Both simulations considered an isolated blade row. Results from these investigations suggest that actuator disk theory provides adequate predictions for conditions with He < 0.6 but becomes inaccurate when describing evanescent modes at the leading and trailing edges and mode distortion of initially planar waves by the mean flow. The advection, dissipation, and distortion of entropy modes were further examined through three-dimensional simulations in a fully developed turbulent channel flow by Morgans et al. (2013) and a transonic turbine stage by Papadogiannis et al. (2015). These simulations emphasized the importance of three-dimensional effects for the transmission of high-frequency temperature perturbations, indicating that low-order models and probably also two-dimensional calculations most likely overpredict the indirect noise generation in turbine stages with multidimensional blade geometries.

4.2. Experimental Analysis

Some of the first experiments to investigate entropy noise in supersonic and subsonic nozzles were performed by Zukoski & Auerbach (1976) and Bohn (1976). In these experiments, electric heaters were installed in the flow path upstream of the nozzle to periodically heat the gas. The induced temperature variations were limited to a few Kelvins. Because the heating introduces acoustic fluctuations due to density variations in addition to entropy perturbations, the interpretation of the measurements was complicated, and compensation techniques were developed in an attempt to separate both contributions.

Bake et al. (2008, 2009b) undertook a comprehensive measurement campaign to systematically investigate entropy noise mechanisms in a canonical flow configuration. To this end, an entropy wave generator facility was constructed. The entropy wave generator was instrumented with thermocouples, a vibrometer, and microphones, and the impedance at the outlet was characterized to establish a data set for analysis and model evaluation. Parametric investigations included the consideration of variations in the mass flow rate, nozzle Mach number, modulation of the heating power, and nozzle geometry. These measurements showed that the pressure fluctuations in the nozzle throat increase monotonically up to Ma $\simeq 0.6$ and decrease for higher Mach numbers. Howe (2010) suggested that this reduction in the overall sound generation can be explained by combined effects of the cancellation of entropy noise by vortex sound due to flow separation and the distortion of temperature inhomogeneities in the nozzle. Analysis by Leyko et al. (2011) suggested that the measured pressure signal contained contributions from acoustic reflections at the exhaust system. Other source mechanisms that contribute to indirect broadband-noise generation have recently been examined by considering the acceleration of vorticity fluctuations (Kings & Bake 2010) and the acceleration of cold spots through a nozzle in a hot acoustic test rig (Knobloch et al. 2015).

Apart from fundamental investigations of individual excitation mechanisms by entropy and vorticity, experiments on generic combustors were conducted to examine the effects of unsteady turbulent combustion on direct and indirect noise emission (Eckstein et al. 2006, Bake et al. 2009a). These generic combustor designs retain essential features of fuel injection, mixing, flame stabilization, and operating conditions that are representative of realistic gas-turbine combustors, yet experimental instrumentation and modular geometry enable independent model evaluation in numerical simulations. Although substantially more challenging, these experiments often reveal physical mechanisms that are not contained in fundamental experiments. For example, Bake et al. (2009a) observed an augmented broadband noise-generation mechanism at frequencies above 1 kHz, which they attributed to the interaction of entropy and vorticity fluctuations with shear layers and turbulent nozzle flow, as well as the strong swirl component introduced by the fuel injector. Eckstein et al. (2006) focused on the generation and feedback of entropy waves with selfexcited flames in a generic liquid-fuel combustor. With relevance to the acoustic nozzle reflection and indirect noise generation, they found that the reflected acoustic waves, π_a^-/ξ_a , are insufficiently strong to alter the thermoacoustic mode. These augmented mechanisms and secondary interactions are typically not considered in idealized configurations or included in theoretical models.

4.3. Discussion and Research Needs

Many of the theoretical models reviewed in previous sections rely on simplifications that limit the consideration of viscous-dissipative effects, boundary layers, flow separation, and the nonlinear interaction of perturbations with turbulence and the mean flow. Global stability analysis and linearized multidimensional Navier-Stokes simulations represent attractive techniques to examine the impact of these effects, thereby bridging the gap between low-order analytic methods and computationally expensive high-fidelity simulations.

The extended theory derived in Section 4.1 suggests that inhomogeneities in the mixture composition represent an additional mechanism for indirect combustion noise. This mechanism exhibits a direct dependence on the combustion regime, operating conditions, and product-gas composition; theoretical and experimental investigations can assist in examining its importance to the overall noise emission.

There is a need for the systematic examination of coupling effects between the interaction of the combustor and downstream engine components. Results from these investigations will guide modeling techniques that often rely on the one-way coupling between engine subcomponents, the specification of boundary conditions, and model selection for predicting unsteady combustion processes. Although the role of entropy noise as a feedback mechanism for thermoacoustic instabilities has been recognized, further experimental studies are needed to quantify the interaction mechanisms of reflected pressure waves and unsteady combustion at conditions relevant for gas-turbine engines. Furthermore, we need to address the lack of detailed multidimensional measurements of noise generation from engine flow paths that include the combustor, turbine, nozzle, jet exhaust, and acoustic far field. Making these experiments viable for model evaluation requires the characterization of boundary conditions, statistical information about velocity and scalar combustion fields, and spectral data to establish causalities between far-field acoustic radiation and noise-generating mechanisms inside the flow path.



Schematic illustration of jet-noise modulation through direct combustion noise and excitation of the shear layer by mean-flow inhomogeneities and perturbations generated upstream of the nozzle.

5. JET-EXHAUST NOISE AND JET-NOISE MODULATIONS

The analysis of core noise is commonly restricted to direct and indirect combustion noise mechanisms that originate upstream of the exhaust nozzle. Previous studies on combustion-generated noise therefore stopped at the nozzle exit, implicitly assuming that nozzle-upstream generated noise by unsteady combustion and internal flow-path interactions directly propagates to the far field. This approach, however, neglects secondary noise mechanisms induced through modifications of the jet-noise sources by mean-flow deformation and perturbations exiting the engine core. In particular, it is known that disturbances, boundary layers, and operating conditions impact the spatiotemporal evolution of the shear layer, potential core, and turbulent flow-field structure of the exhaust jet and acoustic sound field. In addition to core-noise contributions, it is therefore pertinent to also consider these induced jet-noise mechanisms in the analysis of engine-core noise.

5.1. Jet-Noise Modulation

Three main mechanisms can be identified as affecting the flow field and noise generation of mixing layers and jets: (*a*) nozzle exit conditions, (*b*) external excitations, and (*c*) operating conditions (**Figure 6**). Significant efforts have been undertaken to investigate the role of initial conditions and forcing on the jet noise (Crighton 1981, Ho & Huerre 1984). It is therefore instructive to examine essential physical processes in an attempt to link these to conditions relevant to engine-core noise.

Pioneering work by Crow & Champagne (1971) on isothermal subsonic jets has shown that the initial thickness and state (laminar/turbulent) of the boundary layer and the fluctuating intensity at the nozzle exit control the development of the flow field and transition. The evolution of the shear layer in jets emanating from nozzles with weakly disturbed exit conditions is governed by linear instability waves (Hussain & Zedan 1978). Initially, axisymmetric large-scale vortex pairing is observed near the nozzle exit, which transitions to jet-column instabilities with azimuthal variations. The rapid growth of the shear layer reduces the length of the potential core, and velocity fluctuations peak at the location of the first vortex merging, after which they decay to reach a steady level. The pairing of shear-layer vortices introduces an additional sound-producing mechanism that is absent for jets with a turbulent inlet state (Bridges & Hussain 1984). A different picture emerges for initially turbulent nozzle exit conditions. This state is common for practical jet flows and is achieved experimentally, for instance, by tripping the boundary layer in the nozzle. For highly disturbed and turbulent initial states, the transition length is reduced, and the shear layer grows at a lower rate. The evolution of the shear layer is controlled by three-dimensional interactions, and the axisymmetric vortex-pairing mode becomes insignificant. Additionally, there is an extension of the potential core and lower peak turbulence levels that approach a quasi-selfpreserving turbulent state with increasing initial disturbance intensity.

Affected by the nozzle-exit conditions, the evolution of the shear layer and turbulence generation in the jet induces different acoustic sources (Mollo-Christensen et al. 1964, Bridges & Hussain 1984, Harper-Bourne 2010). From an analysis of jet-noise data, Tam et al. (1996, 2008) deduced two empirical spectra that match the radiation characteristics in downstream and sideline directions. Wave-packet analysis, recently reviewed by Jordan & Colonius (2013), provides a physics-based description of jet-noise mechanisms. In this analysis, the coherent portion of the velocity field of a turbulent jet is represented by a wave packet, which is characterized by a constant convection velocity, low azimuthal wave number, and spatial coherence over scales larger than the integral scales. Because of this extended coherence, this wave packet is acoustically more efficient and is linked to the sound radiation through the induced near-field pressure.

Because vortex-pairing sound and higher turbulence levels are observed in jets with laminar or weakly disturbed inlet conditions, they are significantly louder than jets in which this large-scale transition route is bypassed by highly disturbed nozzle-exit conditions. Experiments on jets at low and high subsonic flow conditions by Zaman (2012) confirmed this behavior, showing that a jet with a tripped boundary layer emits less noise. These measurements, however, also indicated that a higher turbulence intensity in the nozzle core increases the noise radiation. Fontaine et al. (2015) performed experiments to examine the effect of the turbulent boundary-layer thickness on the jet noise, showing that the downstream radiation of the high-frequency sound scales with the momentum thickness at the nozzle exit, while the spectral peak was less affected by the boundary-layer state. In contrast, the frequency of the sideline radiation showed better collapse when scaling with the nozzle diameter, suggesting that distributed small-scale turbulence in the jet is the prevailing noise source.

Bogey et al. (2012) and Bogey & Marsden (2013) performed LES calculations in which the initial boundary-layer state was systematically modulated, showing a dramatic reduction of the overall sound pressure level, in excess of 12 dB, at the 90°-sideline angle between untripped and tripped boundary layers at peak turbulence levels of 12%. This difference decreases slightly with increasing jet-forward direction. The persistence of weak vortex-pairing noise was attributed to moderate–Reynolds number conditions and shear-layer thickness, suggesting that large-scale structures remain present even at highly turbulent nozzle-exit conditions.

The noise radiation from heated jets was first investigated by Hoch et al. (1973) and Tanna (1977), and the work of Viswanathan (2004) provides an excellent overview of current knowledge.

Although several questions regarding density effects and noise-source mechanisms in heated jets remain open, it is now understood that the observed change in the sound spectrum and the occurrence of a hump near the spectral peak at increasing temperature are direct results of the reduced Reynolds number. Previously reported shifts in the peak frequency and increased noise levels at higher frequencies are now linked to parasitic noise originating from the test rig. This suggests that these spurious jet-noise contaminations can become increasingly important for engines with low exit Mach number (due to increasing bypass ratio) and significant exhaust enthalpy.

Apart from the linear propagation of internally generated noise to the far field, acoustic fluctuations contribute to the broadband amplification of jet noise through tonal excitation. Bechert & Pfizenmaier (1975, 1977) and Moore (1977) were the first to demonstrate that tonal excitation can lead to a substantial increase in the sound pressure, reaching values in excess of 7 dB. Moore (1977) showed that broadband noise amplification depends on the level and amplitude of the tonal excitation and jet Mach number. The amplification reduces with increasing Mach number, and the directivity remains largely unaffected by the forcing. A maximum level of amplification is achieved at forcing frequencies around the preferred Strouhal number, and substantial amplification can also be achieved through broadband forcing. This was attributed to the excitation of preferredmode instabilities. Excitation at low-amplitude levels resulted in locking with the linear modes, and a nonlinear response in conjunction with modifications of the downstream development of turbulence was observed at higher excitation levels.

In contrast to these investigations, measurements by Kibens (1980), Laufer & Yen (1983), and Hussain & Hasan (1985) on jets at low–Mach number and low–Reynolds number conditions showed a reduction of the broadband noise and excitation of subharmonics when forcing at the shear-layer instability frequency. At these conditions, the harmonic forcing alters the large-scale shear-layer structure, and the organized vortex pairing introduces a stationary sound source with strong directivity. Although these forcing frequencies belong to the upper range of core noise, it is important to recognize that these operating conditions are quite relevant for turboshaft engines in auxiliary power units or helicopters and compact high-power-density engine cores.

Few studies have examined the effect of excitation on heated jets. Compared to unheated jets, Jubelin (1980) reported that broadband amplification in heated jets is reduced and substantially more localized around the excitation frequency. The amplification is less uniform and most prominent toward the forward arc. Furthermore, heated jets are substantially more receptive to upstream excitations, but their receptivity decreases with increasing temperature, pointing at effects of higher dissipation and shear-layer laminarization due to increased temperature-dependent viscosity. O'Brien et al. (2016) performed simulations of a representative engine-core flow path, consisting of independently verifiable components, by using a coupled modeling approach to examine effects of nozzle-upstream perturbations on far-field noise. In this approach, they simulated a dual-swirl gas-turbine combustor using compressible reacting LES; represented the transfer function of a single-stage turbine by actuator disk theory; obtained the flow through the nozzle and in the jet from compressible LES, matching test point 49 of Tanna (1977) with a jet-exit Reynolds number of Re_I = 2.3×10^5 and temperature ratio of $T_{\rm I}/T_{\infty} = 2.857$; and evaluated the acoustic far field using the method due to Ffowcs Williams & Hawkings (1969). Compressible reacting LES are necessary to represent pressure fluctuations inside the combustion chamber. Although these calculations are computationally expensive, advances in high-performance computing and computational resources have made such calculations feasible. In addition, progress in the formulation of predictive combustion models, boundary conditions, and wall models and improved resolution of the turbulent flow field in the combustor, nozzle, and jet exhaust allow one to represent acoustic spectra and flow dynamics over the frequency range of practical relevance.



Figure 7

(*a*) Hybrid simulation of a representative flow path, consisting of independently verifiable engine components, combining compressible reacting large-eddy simulation (LES) of a dual-swirl gas-turbine combustor, actuator disk theory of a single-stage turbine, compressible LES of nozzle and turbulent jet flow, and Ffowcs Williams & Hawkings's (1969) method for the far-field noise radiation. (*b*) Acoustic spectra and (*c*) directivity at a distance of $d/D_{\rm I} = 72$.

Figure 7 shows results for the flow path components of the combustor, turbine, nozzle, and jet exhaust and for the acoustic far field. Predictions of the acoustic spectra and directivity show that far-field noise amplification by direct and indirect combustion noise is confined to downstream radiation angles of $\phi \leq 70^{\circ}$ with a maximum noise amplification of 3 dB compared to undisturbed (clean) nozzle flow. A mild attenuation of far-field noise is observed at sideline and upstream radiation angles, which is attributed to noise reduction at Strouhal numbers above 0.2 and reduced contributions from core noise at these radiation angles. These simulation results indicate that the nozzle and jet-exhaust flow require consideration in the analysis of engine-core noise.

5.2. Discussion and Research Needs

The following physical description of the effects of nozzle-upstream perturbations and mean-flow inhomogeneities on the far-field noise radiation emerges (see **Figure 6**). Owing to scale separation, low-frequency acoustic waves (with $St_J \lesssim 0.1$) generated upstream of the nozzle are directly

transmitted to the far field without significant interaction with the shear layer or turbulent flow field. In contrast, tonal and broadband acoustic and velocity perturbations at higher frequencies (St_J $\gtrsim 0.2$) can couple with the initial shear layer and jet-column instability. Hydrodynamic disturbances from turbulence, thermal, and compositional fluctuations convected by the mean flow are additional sources for triggering instabilities. Depending on the flow conditions and initial state of the shear layer, the linear response to small perturbations at Re_J $\lesssim 10^5$ suppresses broadband noise by containing the energy in large-scale structures. A nonlinear response by large perturbations at Re_J $\gtrsim 10^5$ causes a faster shear-layer transition and higher levels of small-scale turbulence, leading to nearly uniform broadband noise amplification.

In summary, hydrodynamic fluctuations and mean-flow deformations, which are generated upstream of the nozzle, represent potentially equally effective mechanisms at modifying the acoustic source-term distribution in the jet. As such, far-field noise emission can be regarded as a combination of low-frequency core noise and jet noise from fine-scale and large-scale turbulence, which is modulated by perturbations and mean-flow inhomogeneities upstream of the nozzle. Although low-order methods and compact theories can be utilized to reasonably estimate lowfrequency contributions, nonlinear techniques or compressible LES are necessary to represent nozzle-upstream flow-field features that are effective in triggering shear-layer and column-mode instabilities.

Recognizing that nozzle-upstream excitations can effectively attenuate broadband noise, opportunities arise to exploit this mechanism to counterbalance low-frequency amplification by combustion noise. Recent progress on the development of adjoint methods (Kim et al. 2014), resolvent norm analysis (Garnaud et al. 2013), and optimal mode-shape and base-flow modifications (Brandt et al. 2011, Nichols & Lele 2011) represents interesting avenues for application to jet-noise reduction.

6. ENGINE-CORE NOISE

The principal understanding, analysis, and modeling of noise contributions from direct and indirect combustion noise, and the transmission and modulation of jet noise, has guided the investigation of engine-core noise in full-scale engines, auxiliary power units, helicopter engines, and combustor component rigs (Reshotko & Karchmer 1980, Hultgren & Miles 2009, Gordon 2015, Schuster et al. 2015, Stout et al. 2015, Tam & Parrish 2015, Livebardon et al. 2016, O'Brien et al. 2016). However, the detailed spatiotemporal characterization of noise-source mechanisms and acoustic transmission is limited, given restricted experimental access, hostile operating conditions, combustion-physical coupling mechanisms, different far-field noise-source contributions, and geometric complexities. Instead, temperature sensors and microphone probes are commonly employed.

In addition to commonly used spectral analysis of single-point pressure measurements, coherence techniques and modal analyses have been employed to identify far-field noise contributions, source locations, and propagation characteristics of combustion noise. These techniques were successfully applied to the analysis of multiple microphone measurements from different turbofan engines—e.g., AVCO-Lycoming YF-102 (Karchmer & Reshotko 1976), Pratt & Whitney JT15D (Reshotko & Karchmer 1980), General Electric CF6-50 (Doyle & Moore 1980), and Honeywell TECH977 (Miles 2009, 2010; Schuster et al. 2015)—separating indirect and direct combustion noise contributions and determining peak frequencies and spectral shapes of core-noise components. Results from these static engine measurements revealed differences in the core-noise peak frequency, varying between 250 and 400 Hz, and the relative spectral contribution of direct over indirect noise. However, given the superposition of different noise contributions, these analysis tools have limitations in separating noise sources, acoustic and hydrodynamic modes, and low-pass filtering can lead to attenuations of certain noise contributions in cross-correlation analysis (Miles 2010).

Schuster et al. (2015) and Gordon (2015) conducted measurements on an extensively instrumented dual-spool turbofan engine to obtain simultaneous temperature and pressure data in the combustor, interturbine duct, and mixer. Modal analysis in the combustor identified frequencydependent acoustic mode shapes, with a dominating planar mode at frequencies below 500 Hz, and counter-rotating helical and higher-order modes observed at successively higher frequencies. Interestingly, the temperature measurements showed fluctuations with a magnitude in excess of 500 K in the combustor, which decayed by an order of magnitude in the interturbine duct. The cross-spectral analysis showed weak spatial coherence of the temperature signal in the combustion chamber, which was attributed to the inhomogeneous combustion and mixing field, and significantly higher coherence in the interturbine duct.

Miles (2009) identified significant differences in the coherence characteristics between real engines and generic single-cup combustor configurations. This was attributed to the presence of azimuthal modes in annual combustors, higher power settings, and the influence of sound sources from the fan, jet, and combustor-turbine interaction. Such effects require consideration in modeling and in relating measurements on idealized configurations to real engine tests. As such, full-scale engine tests fill an important gap in analyzing noise-source mechanisms, calibrating low-order models, and identifying global performance quantities.

Tam & Parrish (2015) used measurements from an F-22A tactical military aircraft to discriminate noise components that are unique to full-scale engines. Comparisons against similarity spectra indicate the presence of a dominant noise component with directivity along shallow jet angles and dependence on engine-power setting. This source was attributed to possible contributions from indirect combustion noise; however, its persistence at high frequencies is different from the results of Miles (2010), who associated indirect combustion noise to frequencies below approximately 500 Hz. Stout et al. (2015) undertook detailed investigations on the same aircraft by using a four-microphone intensity probe. Results from this analysis showed that the jet-source region contributing to the maximum acoustic intensity moves upstream with increasing frequency, and the acoustic source location moves downstream and broadens for conditions in which the afterburner is employed.

The development and selection of models for the prediction of engine-core noise are dictated by user-specific requirements about accuracy, fidelity, and computational cost. Different approaches have been pursued. Modeling approaches for predicting noise and flow-field dynamics in the combustor, turbine/nozzle, and jet can be differentiated into low-order approaches and multidimensional numerical simulations.

At the lowest level of computational complexity are low-order empirical combustion-noise models for far-field directivity and spectral distribution. These models often rely on semi-empirical information for total acoustic power that are developed with information from engine-test data. They are calibrated by engine manufactures for specific engine types (Stone et al. 2011, Hultgren 2012) and take into account information about combustor geometry, operating conditions, combustion environments, and turbine losses. These semi-empirical combustion-noise models are utilized in aircraft noise prediction programs (ANOPP) (Zorumski 1982). Representative results are shown in **Figure 8***a*, comparing component noise predictions and far-field measurements of a TECH977 turbofan engine at 60% corrected fan speed at two polar radiation angles measured with respect to the downstream direction (Royalty & Schuster 2008). These models provide adequate descriptions of noise levels for engine classes used in their development. However, their applicability to new engine designs as well as outside of their calibration range is uncertain.



Figure 8

Comparison of different modeling approaches for the prediction of engine-core noise. (*a*) Results from a low-order aircraft noise prediction program (ANOPP) for a TECH977 dual-spool high-bypass-ratio turbofan. (*b*) Hybrid simulations of an engine sector of a turboshaft helicopter engine combining large-eddy simulation (LES) for the combustor simulation and actuator disk theory for the turbine. Results adapted from Royalty & Schuster (2008) and Livebardon et al. (2016) with permission from the American Institute of Aeronautics and Astronautics and Elsevier.

In contrast to low-order models, unsteady three-dimensional simulations represent the other spectrum of core-noise predictions. In particular, LES methods are particularly attractive for accurately describing turbulent flows and unsteady source mechanisms that contribute to the generation and transmission of engine-core noise. As such, LES techniques have been employed successfully to predict turbulent combustion in generic and realistic gas-turbine combustors and jet-exhaust flows. However, because of the computational complexity, consideration of moving geometries, and disparity of timescales associated with subsonic combustion and transmic turbine and jet-exhaust flows, the monolithic simulation of the entire engine core using a single solver has not yet been considered.

Coupled methods have been developed for the computationally efficient prediction of engine flow paths (Schlüter et al. 2005). In these methods, multiple specialized flow solvers are combined to enable accurate and efficient predictions of integrated engine components. An example of a coupled method applied to a representative engine flow path is illustrated in **Figure 7**, and other techniques can be considered (Tyacke & Tucker 2015). The main distinguishing features are in the selection of submodels employed for simulating individual engine components, coupling and exchange of information between solvers, and synchronization of different solvers.

LES is an attractive method for simulating flows in the combustor because it facilitates the consideration of multicomponent evaporation, mixing, turbulence, chemical reaction, heat release, the generation of direct combustion noise, and potentially also the occurrence of thermoacoustic instabilities. The representation of the flow field and sound in the turbine allows for different techniques. Analytic methods, discussed in Section 4, provide estimates for the acoustic transfer matrix across turbine stages. However, these simplified methods rely on assumptions about the geometry and mean-flow representation that limit their predictive characteristics. Multidimensional simulations overcome these issues, and Reynolds-averaged Navier-Stokes (RANS) methods are commonly employed to represent the ensemble-averaged turbulent flow field, boundary layers, and rotor/stator wake structures. Solutions from these RANS calculations can then be used as input to acoustic analogies, multiple-scale techniques, or linear methods for the computation of the acoustic field. Alternatively, compressible nonreacting LES can be employed to obtain the unsteady flow field through the turbine. However, the high Reynolds numbers and requirements for resolving the boundary layer, tip gaps, and rotating geometry introduce substantial resource requirements that impose limitations for implementation in routine applications. To accurately capture the interaction of different engine components, such as hot-streak propagation, acoustic reflections, and instabilities, one must couple different solution techniques. Because different solution methods solve for different flow-field quantities (primitive, conserved, and characteristic quantities), employ different decomposition methods (filtering, ensemble averaging, harmonic methods), and use different spatiotemporal resolutions, prolongation and restriction operators are necessary to consistently exchange information across interfaces. A mathematically stable and physically consistent interface is particularly important for core-noise simulations to eliminate spurious acoustic reflections or dispersion of acoustic waves.

Livebardon et al. (2016) used a coupled method combining compressible reacting LES for a sector of the combustor and actuator disk theory for the turbine to simulate combustion noise of a helicopter engine. The propagation of the core noise to the far field and coupling with the exhaust jet were not included. This investigation focused on examining contributions of direct and indirect combustion noise. Results from this study are illustrated in **Figure 8***b*, showing comparisons of measured power spectral density and computations of indirect combustion noise at the turbine exit. Interestingly, these simulations showed that indirect noise is the main contributor to turbine-exit noise, and the low level of direct noise was attributed to the consideration of a single-sector combustion chamber.

FUTURE ISSUES

- 1. The physical description and computational modeling techniques for the acoustic sources from unsteady turbulent combustion are not complete. In particular, the impact of localized and stochastic combustion events associated with ignition, extinction, and transient blowout dynamics on noise radiation and the generation of acoustic precursor events requires further analysis. Understanding these events becomes increasingly important with the implementation of compact combustion chambers, premixed combustion strategies, and high-power engine cores. In addition, advances in the development of high-fidelity combustion models and subgrid closures are necessary to accurately predict these unsteady combustion events and acoustic sources. DNS and high-speed diagnostics can have a significant impact in addressing these issues.
- 2. Further fundamental examination of the coupling of engine components and the characterization of the impact of acoustic reflection on the onset of instabilities are needed. This has direct consequences on the modeling of thermoacoustic precursor events, the formulation of boundary conditions in numerical simulations, and the integration of coupled engine-core noise models.

- 3. The relative contribution of direct and indirect combustion noise is strongly dependent on the engine type, operating conditions, and interaction with other noise-source mechanisms. Further research is therefore needed to describe these contributions in integrated engine flow paths under the consideration of interaction processes between the combustor, turbine, nozzle, and jet exhaust.
- 4. Other noise sources, such as those arising from jet-noise modulation, turbine tones, and thermoacoustic instabilities, need further study to ensure that noise goals are met.
- 5. The description of the generation of indirect noise and the transmission and attenuation of direct combustion noise in engine cores largely relies on low-order models. Further progress is needed to improve model fidelity to include geometric complexities, realistic flow-field descriptions, and additional contributions from hydrodynamic and thermochemical perturbations through nonlinear coupling between turbulence, boundary layers, and mean flow.
- 6. The need for full-scale engine tests is recognized for evaluating design modifications and validating noise-prediction models. Multidimensional flow-field measurements in the combustor and turbine and the characterization of boundary conditions are desirable to effectively use these measurements to validate high-fidelity models.
- 7. To address the pressing need for meeting ambitious noise-reduction goals, opportunities arise for suppressing combustion noise at the source through combustor-design modifications. In addition, active and passive control of nozzle-upstream perturbations and the mean flow are attractive pathways for far-field noise reduction by jet-noise modulations. Adjoint methods and base-flow modifications represent interesting techniques to guide the development of such noise-mitigation strategies.

DISCLOSURE STATEMENT

The author is not aware of any biases that might be perceived as affecting the objectivity of this review.

ACKNOWLEDGMENTS

Financial support through the National Science Foundation and the Office of Naval Research is gratefully acknowledged. I am thankful to the following students and colleagues who provided results, data, and valuable comments on this article: Friedrich Bake, Lennart Hultgren, Jeonglae Kim, Sanjiva Lele, Luca Magri, Ramani Mani, and Jeff O'Brien.

LITERATURE CITED

- Atvars J, Schubert LK, Grande E, Ribner HS. 1966. *Refraction of sound by jet flow or jet temperature*. Tech. Rep. NASA CR-494, NASA, Washington, DC
- Bailly C, Bogey C, Candel S. 2010. Modelling of sound generation by turbulent reacting flows. *Int. J. Aeroacoust*. 9:461–90
- Bake F, Kings N, Fischer A, Röhle I. 2009a. Experimental investigation of the entropy noise mechanism in aero-engines. *Int. J. Aeroacoust.* 8:125–42
- Bake F, Kings N, Röhle I. 2008. Fundamental mechanism of entropy noise in aero-engines: experimental investigations. J. Eng. Gas Turbines Power 130:011202

- Bake F, Richter C, Mühlbauer C, Kings N, Röhle I, et al. 2009b. The entropy wave generator (EWG): a reference case on entropy noise. *J. Sound Vib.* 326:574–98
- Bechert D, Pfizenmaier E. 1975. On the amplification of broad band jet noise by a pure tone excitation. J. Sound Vib. 43:581–87
- Bechert D, Pfizenmaier E. 1977. Amplification of jet noise by higher-mode acoustic excitation. AIAA J. 15:1268–71
- Blanes S, Casas F, Oteo JA, Ros J. 2009. The Magnus expansion and some of its applications. *Phys. Rep.* 470:151–238
- Bloy AW. 1979. The pressure waves produced by the convection of temperature disturbances in high subsonic nozzle flows. 7. Fluid Mech. 94:465–75
- Bogey C, Marsden O. 2013. Identification of the effects of the nozzle-exit boundary-layer thickness and its corresponding Reynolds number in initially highly disturbed subsonic jets. *Phys. Fluids* 25:055106
- Bogey C, Marsden O, Bailly C. 2012. Influence of initial turbulence level on the flow and sound fields of a subsonic jet at a diameter-based Reynolds number of 10⁵. *J. Fluid Mecb.* 701:352–85
- Bohn MS. 1976. Noise production by the interaction of acoustic waves and entropy waves with high-speed nozzle flows. PhD Thesis, Calif. Inst. Technol., Pasadena
- Bragg SL. 1963. Combustion noise. J. Inst. Fuel 36:12-16
- Brandt L, Sipp D, Pralits JO, Marquet O. 2011. Effect of base-flow variations in noise amplifiers: the flat-plate boundary layer. J. Fluid Mech. 687:503–28
- Bridges JE, Hussain AKMF. 1984. Roles of initial condition and vortex pairing in jet noise. J. Sound Vib. 117:289-311
- Bui TP, Ihme M, Schröder W, Pitsch H. 2009. Analysis of different sound source formulations to simulate combustion generated noise using a hybrid LES/APE-RF method. Int. J. Aeroacoust. 8:95–124
- Burnley VS, Culick FEC. 2000. Influence of random excitations on acoustic instabilities in combustion chambers. *AIAA J.* 38:1403–10
- Bushell KW. 1971. A survey of low velocity and coaxial jet noise with application to prediction. J. Sound Vib. 17:271–82
- Candel S. 1972. Analytical studies of some acoustic problems of jet engines. PhD Thesis, Calif. Inst. Technol., Pasadena
- Candel S. 2002. Combustion dynamics and control: progress and challenges. Proc. Combust. Inst. 29:1-28
- Candel S, Durox D, Ducruix S, Birbaud AL, Noiray N, Schuller T. 2009. Flame dynamics and combustion noise: progress and challenges. *Int. J. Aeroacoust.* 8:1–56
- Chang CT, Lee CM, Herbon JT, Kramer SK. 2013. NASA environmentally responsible aviation project develops next-generation low-emissions combustor technologies (Phase I). *J. Aeronaut. Aerosp. Eng.* 2:116
- Chiu HH, Summerfield M. 1974. Theory of combustion noise. Acta Astronaut. 1:967-84
- Chu BT, Kovásznay LSG. 1958. Non-linear interactions in a viscous heat-conducting compressible gas. *J. Fluid Mecb.* 3:494–514
- Colin O, Ducros F, Veynante D, Poinsot T. 2000. A thickened flame model for large eddy simulations of turbulent premixed combustion. *Phys. Fluids* 12:1843–63
- Crighton DG. 1981. Acoustics as a branch of fluid mechanics. J. Fluid Mech. 106:261-98
- Crighton DG. 1992. Computational aeroacoustics for low Mach number flows. In *Computational Aeroacoustics*, ed. JC Hardin, MY Hussaini, pp. 50–68. New York: Springer
- Crighton DG, Dowling AP, Ffowcs Williams JE, Heckl M, Leppington FG. 1992. Modern Methods in Analytical Acoustics: Lecture Notes. London: Springer
- Crow SC, Champagne FH. 1971. Orderly structure in jet turbulence. J. Fluid Mech. 48:547-91
- Culick FEC. 2006. Unsteady Motions in Combustion Chambers for Propulsion Systems. Brussels: NATO
- Cumpsty NA, Marble FE. 1977a. Core noise from gas turbine exhausts. J. Sound Vib. 54:297-309
- Cumpsty NA, Marble FE. 1977b. The interaction of entropy fluctuations with turbine blade rows; a mechanism of turbojet engine noise. *Proc. R. Soc. Lond. A* 357:323–44
- Doak PE. 1972. Analysis of internally generated sound in continuous materials: 2. A critical review of the conceptual adequacy and physical scope of existing theories of aerodynamic noise, with special reference to supersonic jet noise. *J. Sound Vib.* 25:263–335

Dowling AP. 1995. The calculation of thermoacoustic oscillations. J. Sound Vib. 180:557-81

Dowling AP, Mahmoudi Y. 2015. Combustion noise. Proc. Combust. Inst. 35:65-100

- Dowling AP, Stow SR. 2003. Acoustic analysis of gas turbine combustors. J. Prop. Power 19:751-64
- Doyle VL, Moore MT. 1980. Core noise investigation of the CF6-50 turbofan engine. Tech. Rep. NASA CR-159749, NASA, Washington, DC
- Duran I, Moreau S. 2013. Solution of the quasi-one-dimensional linearized Euler equations using flow invariants and the Magnus expansion. J. Fluid Mech. 723:190–231
- Duran I, Morgans AS. 2015. On the reflection and transmission of circumferential waves through nozzles. 7. Fluid Mecb. 773:137–53
- Eckstein J, Freitag E, Hirsch C, Sattelmayer T. 2006. Experimental study on the role of entropy waves in low-frequency oscillations in a RQL combustor. *J. Eng. Gas Turbines Power* 128:264–70
- Ewert R, Schröder W. 2003. Acoustic perturbation equations based on flow decomposition via source filtering. J. Comput. Phys. 188:365–98
- Ffowcs Williams JE, Hawkings DL. 1969. Sound generation by turbulence and surfaces in arbitrary motion. *Philos. Trans. R. Soc. Lond. A* 264:321–42
- Flemming F, Sadiki A, Janicka J. 2007. Investigation of combustion noise using a LES/CAA hybrid approach. Proc. Combust. Inst. 31:3189–96
- Fontaine RA, Elliott GS, Austin JM, Freund JB. 2015. Very near-nozzle shear-layer turbulence and jet noise. J. Fluid Mecb. 770:27–51
- Garnaud X, Lesshafft L, Schmid PJ, Huerre P. 2013. The preferred mode of incompressible jets: linear frequency response analysis. *J. Fluid Mech.* 716:189–202
- Giauque A, Huet M, Clero F. 2012. Analytical analysis of indirect combustion noise in subcritical nozzles. J. Eng. Gas Turbines Power 134:111202
- Gicquel O, Darabiha N, Thevenin D. 2000. Laminar premixed hydrogen/air counterflow flame simulations using flame prolongation of ILDM with differential diffusion. *Proc. Combust. Inst.* 28:1901–8
- Goh CS, Morgans AS. 2011. Phase prediction of the response of choked nozzles to entropy and acoustic disturbances. J. Sound Vib. 330:5184–98
- Goldstein ME. 1976. Aeroacoustics. New York: McGraw-Hill
- Goldstein ME. 2003. A generalized acoustic analogy. 7. Fluid Mech. 488:315-33
- Gordon G. 2015. Acoustic database for turbofan engine core-noise sources, volume I—final report. Tech. Rep. NASA/CR-2015-218879/VOL1, NASA Glenn Res. Cent., Cleveland, OH
- Hardin JC, Pope DS. 1994. An acoustic/viscous splitting technique for computational aeroacoustics. *Theor. Comput. Fluid Dyn.* 6:323–40
- Harper-Bourne M. 2010. Jet noise measurements: past and present. Int. J. Aeroacoust. 9:559-88
- Hassan HA. 1974. Scaling of combustion-generated noise. J. Fluid Mech. 66:445-53
- Haworth DC. 2010. Progress in probability density function methods for turbulent reacting flows. Prog. Energy Combust. Sci. 36:168–259
- Ho CM, Huerre P. 1984. Perturbed free shear layers. Annu. Rev. Fluid Mech. 16:365-424
- Hoch RG, Duponchel JP, Cocking BJ, Bryce WD. 1973. Studies of the influence of density on jet noise. J. Sound Vib. 28:649–68
- Howe MS. 2010. Indirect combustion noise. J. Fluid Mech. 659:267-88
- Hubbard HH, Lassiter LW. 1953. Experimental studies of jet noise. J. Acoust. Soc. Am. 25:381-84
- Huet M, Giauque A. 2013. A nonlinear model for indirect combustion noise through a compact nozzle. J. Fluid Mecb. 733:268–301
- Hultgren LS. 2011. Core noise: implications of emerging N+3 designs and acoustic technology needs. Tech. Rep. E-17796, Acoust. Tech. Working Group, NASA Glenn Res. Cent., Cleveland, OH
- Hultgren LS. 2012. A comparison of combustion-noise models. Presented at AIAA/CEAS Aeroacoust. Conf., 18th, Colorado Springs, CO, AIAA Pap. 2012-2087
- Hultgren LS, Miles JH. 2009. Noise-source separation using internal and far-field sensors for a full-scale turbofan engine. Presented at AIAA/CEAS Aeroacoust. Conf., 15th, Miami, FL, AIAA Pap. 2009-3220
- Hurle IR, Price RB, Sugden TM, Thomas A. 1968. Sound emission from open turbulent premixed flames. Proc. R. Soc. Lond. A 303:409–27

- Hussain AKMF, Hasan MAZ. 1985. Turbulence suppression in free turbulent shear flows under controlled excitation. Part 2. Jet-noise reduction. J. Fluid Mech. 150:159–68
- Hussain AKMF, Zedan MF. 1978. Effects of the initial condition on the axisymmetric free shear layer: effects of the initial momentum thickness. *Phys. Fluids* 21:1100–12
- ICAO (Int. Civil Aviat. Organ.). 2013. Global air transport outlook to 2030 and trends to 2040. ICAO Circ. 333, Int. Civi Aviat. Organ., Quebec
- ICAO (Int. Civil Aviat. Organ.). 2014. NoiseDB database. Noise Certification Database, accessed in December 2015, Int. Civil Aviat. Organ., Quebec. http://noisedb.stac.aviation-civile.gouv.fr
- Ihme M, Cha CM, Pitsch H. 2005. Prediction of local extinction and re-ignition effects in non-premixed turbulent combustion using a flamelet/progress variable approach. Proc. Combust. Inst. 30:793–800
- Ihme M, Kaltenbacher M, Pitsch H. 2006. Numerical simulation of flow- and combustion-induced sound using a hybrid LES/CAA approach. Proc. 2006 Summer Prog., pp. 497–510. Stanford, CA: Cent. Turbul. Res.
- Ihme M, Pitsch H. 2012. On the generation of direct combustion noise in turbulent non-premixed flames. Int. J. Aeroacoust. 11:25–78
- Ihme M, Pitsch H, Bodony D. 2009. Radiation of noise in turbulent non-premixed flames. Proc. Combust. Inst. 32:1545–53
- Job G, Herrmann F. 2006. Chemical potential-a quantity in search of recognition. Eur. J. Phys. 27:353-71
- Jordan P, Colonius T. 2013. Wave packets and turbulent jet noise. Annu. Rev. Fluid Mech. 45:173-95
- Jubelin B. 1980. New experimental studies on jet noise amplification. Presented at AIAA Aeroacoust. Conf., 6th, Hartford, CT, AIAA Pap. 1980-0961
- Karchmer A, Reshotko M. 1976. Core noise source diagnostics on a turbofan engine using correlation and coherence techniques. Tech. Rep. NASA-TM-X-73535, NASA, Washington, DC
- Kerstein AR. 1988. A linear-eddy model of turbulent scalar transport and mixing. *Combust. Sci. Technol.* 60:391-421
- Kibens V. 1980. Discrete noise spectrum generated by an acoustically excited jet. AIAA J. 18:434-41
- Kilham JK, Kirmani N. 1979. The effect of turbulence on premixed flame noise. Proc. Combust. Inst. 17:327-36
- Kim J, Bodony DJ, Freund JB. 2014. Adjoint-based control of loud events in a turbulent jet. *J. Fluid Mech.* 741:28–59
- Kings N, Bake F. 2010. Indirect combustion noise: noise generation by accelerated vorticity in a nozzle flow. Int. J. Spray Combust. Dyn. 2:253–66
- Klein SA, Kok JBW. 1999. Sound generation by turbulent non-premixed flames. *Combust. Sci. Technol.* 149:267–95
- Klimenko AY, Bilger RW. 1999. Conditional moment closure for turbulent combustion. Prog. Energy Combust. Sci. 25:595–687
- Knobloch L, Werner T, Bake F. 2015. Entropy noise generation and reduction in a heated nozzle flow. Presented at AIAA/CEAS Aeroacoust. Conf., 21st, Dallas, TX, AIAA Pap. 2015-2818
- Kotake S. 1975. On combustion noise related to chemical reactions. J. Sound Vib. 42:399-410
- Kotake S, Takamoto K. 1987. Combustion noise: effects of the shape and size of burner nozzle. J. Sound Vib. 112:345–54
- Kumar RN. 1976. Further experimental results on the structure and acoustics of turbulent jet flames. Presented at AIAA Aeroacoust. Conf., 2nd, Hampton, VA, AIAA Pap. 1975-523
- Laufer J, Yen TC. 1983. Noise generation by a low-Mach-number jet. J. Fluid Mech. 134:1-31
- Leyko M, Moreau S, Nicoud F, Poinsot T. 2010. Waves transmission and generation in turbine stages in a combustion-noise framework. Presented at AIAA/CEAS Aeroacoust. Conf., 16th, Stockholm, AIAA Pap. 2010-4032
- Leyko M, Moreau S, Nicoud F, Poinsot T. 2011. Numerical and analytical modelling of entropy noise in a supersonic nozzle with a shock. J. Sound Vib. 330:3944–58
- Leyko M, Nicoud F, Poinsot T. 2009. Comparison of direct and indirect combustion noise mechanisms in a model combustor. AIAA J. 47:2709–16
- Lieuwen TC. 2012. Unsteady Combustor Physics. Cambridge, UK: Cambridge Univ. Press
- Lighthill MJ. 1952. On sound generated aerodynamically. I: General theory. Proc. R. Soc. Lond. A 211:564-87

- Lilley GM. 1974. On the noise from jets. Rep. AGARD-CP-131, Advis. Group Aerosp. Res. Dev., Washington, DC
- Livebardon T, Moreau S, Gicquel L, Poinsot T, Bouty E. 2016. Combining LES of combustion chamber and an actuator disk theory to predict combustion noise in a helicopter engine. *Combust. Flame* 165:272–87
- Magri L, See YC, Tammisola O, Ihme M, Juniper MP. 2017. Multiple-scale thermo-acoustic stability analysis of a coaxial jet combustor. *Proc. Combust. Inst.* 36: In press
- Mahan JR, Karchmer A. 1995. Combustion and core noise. In Aeroacoustics of Flight Vehicles, Theory and Practice, Vol. 1: Noise Sources, ed. HH Hubbard, pp. 483–517. Melville, NY: Acoust. Soc. Am.

Mani R. 1981. Low-frequency sound propagation in a quasi-one-dimensional flow. 7. Fluid Mech. 104:81-92

- Marble FE, Candel SM. 1977. Acoustic disturbance from gas non-uniformities convected through a nozzle. 7. Sound Vib. 55:225–43
- Meier W, Barlow RS, Chen YL, Chen JY. 2000. Raman/Rayleigh/LIF measurements in a turbulent CH₄/H₂/N₂ jet diffusion flame: experimental techniques and turbulence-chemistry interaction. *Combust. Flame* 123:326–43
- Miles JH. 2009. Time delay analysis of turbofan engine direct and indirect combustion noise sources. J. Propuls. Power 25:218–27
- Miles JH. 2010. Separating direct and indirect turbofan engine combustion noise using the correlation function. J. Propuls. Power 26:1144–52
- Mishra A, Bodony DJ. 2013. Evaluation of actuator disk theory for predicting indirect combustion noise. J. Sound Vib. 332:821–38
- Moase WH, Brear MJ, Manzie C. 2007. The forced response of choked nozzles and supersonic diffusers. *J. Fluid Mecb.* 585:281–304
- Mollo-Christensen E, Kolpin MA, Martuccelli JR. 1964. Experiments on jet flows and jet noise far-field spectra and directivity patterns. J. Fluid Mech. 18:285–301
- Mongeau L, Huff D, Tester B. 2013. Aircraft noise technology review and medium and long term noise reduction goals. Proc. Meet. Acoust. 19:040041
- Moore CJ. 1977. The role of shear-layer instability waves in jet exhaust noise. J. Fluid Mech. 80:321-67
- Morfey CL. 1973. Amplification of aerodynamic noise by convective flow inhomogeneities. J. Sound Vib. 31:391–97
- Morfey CL. 2003. The role of viscosity in aerodynamic sound generation. Int. J. Aeroacoust. 2:225-40
- Morgans AS, Goh CS, Dahan JA. 2013. The dissipation and shear dispersion of entropy waves in combustor thermoacoustics. J. Fluid Mecb. 733:R2
- Mühlbauer B, Noll B, Aigner M. 2009. Numerical investigation of the fundamental mechanism for entropy noise generation in aero-engines. Acta Acust. United Acust. 95:470–78
- Muthukrishnan M, Strahle WC, Neale DH. 1978. Separation of hydrodynamic, entropy, and combustion noise in a gas turbine combustor. *AIAA J*. 16:320–27
- Nichols JW, Lele SK. 2011. Non-normal global modes of high-speed jets. Int. J. Spray Combust. Dyn. 3:285– 302
- Obermeier F. 1985. Aerodynamic sound generation caused by viscous processes. J. Sound Vib. 99:111-20
- O'Brien J, Kim J, Ihme M. 2016. Investigation of the mechanisms of jet-engine core noise using large-eddy simulation. Presented at AIAA Aerosp. Sci. Meet., 54th, San Diego, CA, AIAA Pap. 2016-0761
- Palies P, Durox D, Schuller T, Candel S. 2011. Acoustic-convective mode conversion in an aerofoil cascade. J. Fluid Mecb. 672:545–69
- Papadogiannis D, Wang G, Moreau S, Duchaine F, Gicquel L, Nicoud F. 2015. Assessment of the indirect combustion noise generated in a transonic high-pressure turbine stage. J. Eng. Gas Turb. Power 138:041503
- Peake N, Parry AB. 2012. Modern challenges facing turbomachinery aeroacoustics. *Annu. Rev. Fluid Mech.* 44:227–48
- Peters N. 1984. Laminar diffusion flamelet models in non-premixed turbulent combustion. Prog. Energy Combust. Sci. 10:319–39
- Peters N. 2000. Turbulent Combustion. Cambridge, UK: Cambridge Univ. Press
- Phillips OM. 1960. On the generation of sound by supersonic turbulent shear layers. *J. Fluid Mech.* 9:1–28
 Pierce CD, Moin P. 2004. Progress-variable approach for large-eddy simulation of non-premixed turbulent combustion. *J. Fluid Mech.* 504:73–97

- Polifke W, Paschereit CO, Döbbeling K. 2001. Constructive and destructive interference of acoustic and entropy waves in a premixed combustor with a choked exit. *Int. J. Acoust. Vib.* 6:135–46
- Pope SB. 1985. PDF methods for turbulent reactive flows. Prog. Energy Combust. Sci. 11:119-92
- Price RB, Hurle IR, Sugden TM. 1969. Optical studies of the generation of noise in turbulent flames. *Proc. Combust. Inst.* 12:1093–102
- Pridmore-Brown DC. 1958. Sound propagation in a fluid flowing through an attenuating duct. *J. Fluid Mecb.* 4:393–406
- Rajaram R, Lieuwen T. 2003. Parametric studies of acoustic radiation from premixed flames. *Combust. Sci. Technol.* 175:2269–98
- Rajaram R, Lieuwen T. 2009. Acoustic radiation from turbulent premixed flames. 7. Fluid Mech. 637:357-85
- Renfro MW, Chaturvedy A, King GB, Laurendeau NM, Kempf A, et al. 2004. Comparison of OH time-series measurements and large-eddy simulations in hydrogen jet flames. *Combust. Flame* 139:142–51
- Reshotko M, Karchmer A. 1980. Core noise measurements from a small, general aviation turbofan engine. Tech. Rep. NASA-TM-81610, NASA Lewis Res. Cent., Cleveland, OH
- Royalty CM, Schuster B. 2008. Noise from a turbofan engine without a fan from the engine validation of noise and emission reduction technology (EVNERT) program. Presented at AIAA/CEAS Aeroacoust. Conf., 14th, Vancouver, AIAA Pap. 2008-2810
- Schlüter JU, Wu X, Kim S, Shankaran S, Alonso JJ, Pitsch H. 2005. A framework for coupling Reynoldsaveraged with large-eddy simulations for gas turbine applications. *J. Fluids Eng.* 127:806–15
- Schneider C, Dreizler A, Janicka J, Hassel EP. 2003. Flow field measurements of stable and locally extinguishing hydrocarbon-fueled jet flames. *Combust. Flame* 135:185–90
- Schuster B, Gordon G, Hultgren LS. 2015. Dynamic temperature and pressure measurements in the core of a propulsion engine. Presented at AIAA/CEAS Aeroacoust. Conf., 21st, Dallas, TX, AIAA Pap. 2015-2819
- Seo JH, Moon YJ. 2006. Linearized perturbed compressible equations for low Mach number aeroacoustics. J. Comput. Phys. 218:702–19
- Shivashankara BN, Strahle WC, Handley JC. 1974. Evaluation of combustion noise scaling laws by an optical technique. *AIAA J.* 13:623–27
- Singh KK, Frankel SH, Gore JP. 2004. Study of spectral noise emissions from standard turbulent nonpremixed flames. *AIAA 7.* 42:931–36
- Singh KK, Zhang C, Gore JP, Mongeau L, Frankel SH. 2005. An experimental study of partially premixed flame sound. *Proc. Combust. Inst.* 30:1707–15
- Smith TJB, Kilham JK. 1963. Noise generation by open turbulent flames. J. Acoust. Soc. Am. 35:715-24
- Stone JR, Krejsa EA, Clark BJ. 2011. Enhanced core noise modeling for turbofan engines. Tech. Rep. NASA/CR-2011-217026, NASA Glenn Res. Cent., Cleveland, OH
- Stout TA, Gee KL, Neilsen TB, Wall AT, James MM. 2015. Source characterization of full-scale jet noise using acoustic intensity. Noise Control Eng. 7. 63:522–36
- Stow SR, Dowling AP, Hynes TP. 2002. Reflection of circumferential modes in a choked nozzle. J. Fluid Mech. 467:215–39
- Strahle WC. 1971. On combustion generated noise. J. Fluid Mech. 49:399-414
- Strahle WC. 1973. Refraction, convection, and diffusion flame effects in combustion-generated noise. Proc. Combust. Inst. 14:527–35
- Strahle WC. 1978. Combustion noise. Prog. Energy Combust. Sci. 4:157-76
- Talei M, Hawkes ER, Brear MJ. 2013. A direct numerical simulation study of frequency and Lewis number effects on sound generation by two-dimensional forced laminar premixed flames. *Proc. Combust. Inst.* 34:1093–100
- Tam CKW, Golebiowski M, Seiner JM. 1996. On the two components of turbulent mixing noise from supersonic jets. Presented at AIAA Aeroacoust. Conf., State College, PA, AIAA Pap. 1996-1716
- Tam CKW, Parrish SA. 2015. Noise of high-performance aircraft at afterburner. J. Sound Vib. 352:103-28
- Tam CKW, Viswanathan K, Ahuja KK, Panda J. 2008. The sources of jet noise: experimental evidence. J. Fluid Mecb. 615:253–92
- Tanna HK. 1977. An experimental study of jet noise part I: turbulent mixing noise. J. Sound Vib. 50:405-28
- Thomas A, Williams GT. 1966. Flame noise: sound emissions from spark-ignited bubbles of combustible gas. Proc. R. Soc. Lond. A 294:449–66

Tsien HS. 1952. The transfer functions of rocket nozzles. J. Am. Rocket Soc. 22:139-43

- Tyacke JC, Tucker PG. 2015. Future use of large eddy simulation in aero-engines. J. Turbomach. 137:081005
- Tyler JM, Sofrin TG. 1962. Axial flow compressor noise studies. SAE Tech. Pap. 620532, SAE Int., Warrendale, PA
- van Oijen JA, de Goey LPH. 2000. Modelling of premixed laminar flames using flamelet-generated manifolds. *Combust. Sci. Technol.* 161:113–37

Veynante D, Vervisch L. 2002. Turbulent combustion modeling. Prog. Energy Combust. Sci. 28:193-266

- Viswanathan K. 2004. Aeroacoustics of hot jet. J. Fluid Mech. 516:39-82
- Wang M, Freund JB, Lele SK. 2006. Computational prediction of flow-generated sound. Annu. Rev. Fluid Mecb. 38:483–512
- Williams FA. 1985. Combustion Theory. Reading, MA: Perseus
- Wu H, See YC, Wang Q, Ihme M. 2015. A Pareto-efficient combustion framework with submodel assignment for predicting complex flame configurations. *Combust. Flame* 162:4208–30
- Zaman KBMQ. 2012. Effect of initial boundary-layer state on subsonic jet noise. AIAA J. 50:1784-95
- Zhao W, Frankel SH. 2001. Numerical simulations of sound radiated from an axisymmetric premixed reacting jet. Phys. Fluids 13:2671–81
- Zhu M, Dowling AP, Bray KNC. 2001. Self-excited oscillations in combustors with spray atomizers. J. Eng. Gas Turb. Power 123:779–86
- Zorumski WE. 1982. Aircraft noise prediction program theoretical manual, part 2. Tech. Rep. NASA-TM-83199-PT-2, NASA, Washington, DC
- Zukoski EE, Auerbach JM. 1976. Experiments concerning the response of supersonic nozzles to fluctuating inlet conditions. *J. Eng. Power* 98:60–64